

A Service of

ZBШ

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Reisinger, Markus; Thomes, Tim Paul

Working Paper Manufacturer collusion: Strategic implications of the channel structure

DICE Discussion Paper, No. 261

Provided in Cooperation with: Düsseldorf Institute for Competition Economics (DICE), Heinrich Heine University Düsseldorf

Suggested Citation: Reisinger, Markus; Thomes, Tim Paul (2017) : Manufacturer collusion: Strategic implications of the channel structure, DICE Discussion Paper, No. 261, ISBN 978-3-86304-260-8, Heinrich Heine University Düsseldorf, Düsseldorf Institute for Competition Economics (DICE), Düsseldorf

This Version is available at: https://hdl.handle.net/10419/162839

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Düsseldorf Institute for Competition Economics

DISCUSSION PAPER

No 261

Manufacturer Collusion: Strategic Implications of the Channel Structure

Markus Reisinger, Tim Paul Thomes

July 2017



d|u|p düsseldorf university press

IMPRINT

DICE DISCUSSION PAPER

Published by

düsseldorf university press (dup) on behalf of Heinrich-Heine-Universität Düsseldorf, Faculty of Economics, Düsseldorf Institute for Competition Economics (DICE), Universitätsstraße 1, 40225 Düsseldorf, Germany www.dice.hhu.de

Editor:

Prof. Dr. Hans-Theo Normann Düsseldorf Institute for Competition Economics (DICE) Phone: +49(0) 211-81-15125, e-mail: <u>normann@dice.hhu.de</u>

DICE DISCUSSION PAPER

All rights reserved. Düsseldorf, Germany, 2017

ISSN 2190-9938 (online) - ISBN 978-3-86304-260-8

The working papers published in the Series constitute work in progress circulated to stimulate discussion and critical comments. Views expressed represent exclusively the authors' own opinions and do not necessarily reflect those of the editor.

Manufacturer Collusion: Strategic Implications of the Channel Structure^{*}

Markus Reisinger^{\dagger} Tim Paul Thomes^{\ddagger}

July 2017

Abstract

We investigate how the structure of the distribution channel affects tacit collusion between manufacturers. When selling through a common retailer, we find—in contrast to the conventional understanding of tacit collusion that firms act to maximize industry profits—that colluding manufacturers strategically induce double marginalization so that retail prices are above the monopoly level. This lowers industry profits but increases the profit share that manufacturers appropriate from the retailer. Comparing common distribution with independent (exclusive) distribution, we show that the latter facilitates collusion. Despite this result, common retailing leads to lower welfare because a common retailer monopolizes the downstream market. For the case of independent retailing, we also demonstrate that contract offers that are observable to the rival retailer are not necessarily beneficial for collusive purposes.

Keywords: tacit collusion; contract observability; common retailing; independent (exclusive) retailing; two-part tariffs; wholesale price contracts

^{*}We thank a Co-Editor and two anonymous referees as well as Stéphane Caprice, Jaques Crémer, Tommy Staahl Gabrielsen, Germain Gaudin, Teis Lunde Lømo, Volker Nocke, Daniel O'Brien, Patrick Rey, Olga Rozanova, Marius Schwartz, Shiva Shekhar, Yaron Yehezkel, participants at the 2015 MaCCI conference in Mannheim, the 2015 Bergen Competition Policy Conference, the 2015 EEA annual congress in Mannheim, the 2016 IOSE conference in Saint Petersburg and seminar participants at the Universities of Toulouse and Hamburg for very helpful comments and suggestions.

[†]Frankfurt School of Finance & Management, Department of Economics, Sonnemannstr. 9-11, 60314 Frankfurt am Main, Germany. E-Mail: m.reisinger@fs.de.

[‡]Universität Düsseldorf, Düsseldorf Institute for Competition Economics (DICE), Universitätsstrasse 1, 40225 Düsseldorf, Germany. E-Mail: thomes@dice.hhu.de.

1 Introduction

Large manufacturers which sell their products through retailers are often long-term competitors. An example is the automobile industry, where in many countries few big producers control a large share of the market over a long time horizon. For example, General Motors, Ford, Chrysler, and Toyota have supplied around 60% of the US market during the last ten years with relatively stable shares (Plunkett 2012).¹ Almost without exception, car producers sell to final consumers via authorized non-integrated dealers.²

A similar pattern applies to the market for beauty products, cosmetics and toiletries. This market is dominated by few established players, such as Estée Lauder, L'Oreal, Procter & Gamble, and Unilever.³ In contrast to the automobile industry, these companies typically follow a strategy of running a strong wholesale business with retailers carrying the products of multiple competing manufacturers.⁴

The existing literature has pointed out that the channel structure has important effects on manufacturers' profits (see, e.g., O'Brien and Shaffer 1997 or Cachon and Kök 2010 on common (multi-brand) retailing, and Bonanno and Vickers 1988 or Rey and Stiglitz 1995 on independent (exclusive) retailing).⁵ This literature takes a static perspective, that is, it examines the strategic implications of different forms of the structure in a one-shot game. It therefore neglects the effects that the distribution channel may have on repeated interaction between manufacturers. Likewise, the literature on long-term cooperation and collusion between manufacturers has ignored channel considerations. Most papers assume that firms sell directly to final consumers, whereas papers explicitly considering the vertical relationship between manufacturers and retailers (e.g., Nocke and White 2007, or Jullien and Rey 2007) focus on vertical integration or the effects of contracts within the channel. However, in most industries (i) dynamic considerations play an important role, implying that manufacturers likely base their pricing decisions not only on current but also on past prices, and (ii) manufacturers sell their products via retailers to final consumers, implying that the channel structure affects pricing decisions.

¹Similar or even bigger numbers can be found in Europe, where domestic producers control a large part of the market in most countries.

 $^{^{2}}$ As reported by Maxton and Wormald (2004), the average number of brands per dealer is around 1.2 in Europe and 2.4 in the US.

³For example, in the haircare segment, which accounts for about 24% of the industry's total revenue (Panteva 2011), L'Oreal and Procter & Gamble together have a market share well above 40% (Trefis Analysis 2011).

⁴As pointed out by Panteva (2011), mass merchandisers, supermarkets, drugstores and department stores account for the lion's share of the industry's sales at the retail level.

⁵We provide an overview of the relevant literature in the next section.

In fact, several empirical studies provide evidence consistent with the conjecture that pricing in many industries is based on long-term competitive dynamics. Studying the automobile industry, Sudhir (2001a) demonstrates that prices in the compact and midsize segment of the US car market during the 1980s are indicative of manufacturer collusion and cannot be explained by short-term competition.⁶ Similar observations can be made in markets where the prevalent channel structure is common retailing. For example, in the soft drink industry, Saltzman et al. (1999) provide detailed evidence that bottlers colluded on the prices charged to supermarkets and general merchandisers during the 1980s and the early 1990s. Likewise, using data on food categories for two suburban retail stores, Sudhir (2001b) finds that manufacturer pricing is consistent with cooperation.

Building on these considerations, the objective of this paper is to examine how the structure of the distribution channel affects the strategic choices of manufacturers aiming to achieve cooperative outcomes. We seek to address the following questions on the channel structure in a dynamic setting. How does the channel structure affect collusive behavior between manufactures? Are cooperation strategies under common retailing fundamentally different from those under independent retailing? How does contract observability affect tacit collusion between competing manufacturers? Which channel structure leads to a higher welfare? Are the findings robust to changes in the contractual form?

To answer these questions, we consider a simple infinitely-repeated game with two singleproduct manufacturers contracting either with a common retailer or with independent (exclusive) retailers. In the baseline model, the contractual form is a two-part tariff.⁷ We focus on tacit collusion between manufacturers and determine contracting decisions in the collusive agreement and the critical discount factors above which collusion can be sustained. For expositional simplicity, we suppose that the retailer (or the retailers) are short-lived. However, we demonstrate that all our results carry over to the case of long-lived retailers. Within this framework, the channel structure has several distinct effects on manufacturers' collusion.

Concerning the first research question, we show that when manufacturers sell through a common retailer, tacit collusion works in a fundamentally different way than in case they sell directly to final consumers. In particular, colluding manufacturers propose contracts with per unit wholesale prices above marginal cost, deliberately accepting *double marginalization*. Thus, the industry profits under collusion are not maximized and even *below* those of the static game. However, in doing so, manufacturers obtain a larger share of the channel profits.

 $^{^{6}}$ Bresnahan (1987) reaches an analogous conclusion when analyzing the pricing behavior in this industry during the 1950s.

⁷In an extension, we analyze linear wholesale price contracts and compare the results.

The intuition underlying this finding lies in the common retailer's opportunity to threaten a manufacturer with rejecting his offer and only selling the rival's product. This effect allows the retailer to pit manufacturers against each other and keep part of the industry profits. We show that when tacitly colluding, manufacturers face a trade-off between maximizing channel profits and mitigating the common retailer's opportunity of rejecting one contract offer. Our analysis reveals that they achieve the latter by raising the collusive wholesale price above the industry profit maximizing level. Setting the collusive wholesale price sufficiently high lowers the common retailer's profit from selling only one manufacturer's product, thereby squeezing the share of the industry profits that can be kept by the retailer. By contrast, when manufacturers compete against each other (as in the static game), each one offers a contract that maximizes the bilateral profit of the manufacturer and the retailer, which avoids double marginalization and maximizes channel profits. Thus, the manufacturers' collusive strategy is to get a larger share of a smaller 'pie'.

Instead, when manufacturers sell through independent (exclusive) retailers, tacit collusion works in a similar way as in the case in which manufacturers sell directly to final consumers. This holds independently of whether or not contracts are observable to the rival retailer. An independent retailer obtains only an offer from the own manufacturer, thereby lacking the opportunity to sell a different product. Therefore, colluding manufacturers set wholesale prices that maximize industry profits.

We then compare the channel structures with respect to their impact on the stability of manufacturer collusion. As explained above, the common retailer can keep part of the industry profits. This leads to higher collusion profits for manufacturers under independent retailing. However, for the same reason, manufacturers can also realize higher static profits and therefore higher profits along the punishment phase with independent retailers.

We demonstrate that the result is nevertheless unambiguous in favor of independent retailing due to a novel effect: defection from collusion implies that the deviant induces the common retailer to reject the offer of the non-deviating manufacturer because the low wholesale price of the deviant makes the rival manufacturer's contract unattractive for the common retailer. This leads to monopolization of the downstream market. Therefore, the ratio between deviation and collusion profits is higher under common retailing than under independent retailing, implying that the incentive to deviate is larger in the former regime. As a consequence, manufacturer collusion can be sustained for a larger range of discount factors if products are sold through independent retailers.

Despite this result, we show that welfare is larger with independent retailers. The reason is that a common retailer sets monopoly prices in the downstream market, whatever the wholesale prices manufacturers have chosen. Even if the discount factor is such that manufacturer collusion is possible under independent retailing but not under common retailing, the effect that downstream competition is eliminated under the latter dominates.

We then turn to the question if contract observability facilitates collusion under independent retailing. Previous studies on static models (e.g., Coughlan 1985, Rey and Stiglitz 1995) find that contract observability gives rise to a strategic effect: manufacturers set higher wholesale prices, resulting in a dampening of the competitive pressure in the retail market and in higher profits. Thus, in a static framework, contract observability is always profitable.

In a dynamic setting, we demonstrate that contract observability imposes two opposing effects on the sustainability of manufacturer collusion. One the one hand, retailers can immediately react to a deviation of the rival manufacturer, which makes collusion more stable. On the other hand, due to the strategic effect, the punishment following a deviation is less severe, leading to a destabilization of collusive agreements. We show that the second effect dominates if products are close substitutes. The intuition is that in this case the retailer of the non-deviating manufacturer is constrained by the high wholesale price and the possibility to react immediately loses importance. As a consequence, we obtain that observable contracts facilitate collusion if competition is low or moderate, whereas private contracts are more suitable for collusive purposes if competition is fierce. This result provides a strategic rationale for secret contracting, which is not encompassed by static models.

Finally, we show that the main insights derived with two-part tariffs carry over to the case with linear wholesale price contracts.⁸ In particular, tacit collusion under common retailing leads to increased wholesale prices which lower channel profits but increase the share manufacturers can keep. Turning to independent retailing, the only difference is that contract observability unambiguously favors collusion. The intuition is that, due to the strategic effect, manufacturers need a lower wholesale price under public contracts to realize identical collusion profits as under private contracts. This makes deviation always more profitable under secrecy of contracts.

Our paper has also interesting implications for antitrust authorities. It demonstrates that a distribution channel of exclusive retailers is more prone to collusion compared to a channel with a common retailer. For example, the "Guidelines on Vertical Restraints" issued by the European Commission state that "(when) most or all of the suppliers apply exclusive distribution this may (...) facilitate collusion."⁹ Our paper provides a rationale for this statement by demonstrating a clear mechanism why this effect occurs. However, this does not imply that independent retailing is anticompetitive relative to common retailing. Therefore, it does not justify to consider the former channel structure more suspiciously.

⁸Linear wholesale price contract are still prominent in many industries due to their simplicity.

⁹See p. 45, European Commission (2010), Guidelines on Vertical Restraints.

The rest of the paper is organized as follows: Section 2 presents the relation to previous literature. Section 3 sets up the model. Section 4 characterizes the equilibrium under common retailing. Section 5 provides the equilibrium analysis in case manufacturers sell through independent retailers for both secret and public contracts. Section 6 studies sustainability of cooperation under the different channel structures and provides a welfare analysis. Section 7 analyzes linear wholesale price contracts, and Section 8 concludes.

2 Relation to the Literature

This paper contributes to the literature on competition in manufacturer-retailer relationships in several aspects. First, it extends the recent and growing literature on collusion in vertical settings. This literature started with Nocke and White (2007) who analyze whether vertical integration can facilitate tacit collusion between manufacturers under two-part tariffs. They show that this is indeed the case because the downstream affiliate of the integrated firm is no longer a potential buyer for a deviating manufacturer.¹⁰ Normann (2009) considers linear upstream contracts and shows that a similar result obtains even with double marginalization. Jullien and Rey (2007) demonstrate in a model with stochastic demand that resale price maintenance reduces profits in a static framework, but helps to facilitate collusion because it simplifies the detection of deviations. Piccolo and Reisinger (2011) analyze exclusive territories and find that they help to sustain collusion if wholesale contracts are observable because this enables retailers to react to deviations. Piccolo and Miklòs-Thal (2012) analyze collusion at the retail level and find that retailers can collude through manufacturers via above-cost wholesale prices and negative fixed fees. Finally, Gilo and Yehezkel (2016) consider collusion between retailers and a common supplier. They show that the presence of a common supplier helps retailers to sustain collusion (that is, vertical collusion is easier to maintain than horizontal collusion) and demonstrate how collusion profits are distributed between the firms as the discount factor varies.¹¹

To the best of our knowledge, our paper is the first one analyzing the effects of different channel structures on manufacturer collusion in a dynamic framework and how changes in the contractual form affect the respective outcomes.¹²

 $^{^{10}}$ Nocke and White (2010) conclude that the ability to collude is stronger, the larger the size of the integrated downstream buyer.

¹¹Schinkel et al. (2008) consider a model in which retailers can sue manufacturers when wholesale prices exceed those of a competitive environment. They demonstrate that allowing only direct retailers to claim damages facilitates collusion because manufacturers can pay them in exchange for not suing.

¹²Gabrielsen (1997) also studies common and independent retailing in an infinitely repeated game. However, in contrast to our analysis, he does not consider the collusive effects of the channel structure.

Our paper also relates to the work analyzing strategic effects of the channel structure in static environments. One strand of this work focuses on a framework with multiple manufacturers and a common retailer. For example, Choi (1991) analyzes wholesale price contracts and distinguishes between the scenarios in which either manufacturers or retailers are the Stackelberg price leader. O'Brien and Shaffer (1997) analyze the effects of non-linear contracts and exclusive dealing on foreclosure incentives. They show that exclusive dealing leads to lower equilibrium profits for manufacturers than contracts allowing the retailer to carry both brands.¹³ Finally, Cachon and Kök (2010) examine the impact of different contractual forms and find that two-part tariffs and quantity-discount contracts may be disadvantageous for manufacturers compared to wholesale price contracts if products are close substitutes.

A second strand of research focuses on competition between multiple manufacturers selling through independent (exclusive) retailers. McGuire and Staelin (1983) demonstrate that independent retailing leads to a dampening of retail competition and can therefore be preferred by manufacturers to vertical integration.¹⁴ Bonanno and Vickers (1988) demonstrate that this is also the case if manufacturers employ two-part tariffs. In a similar vein, Rey and Stiglitz (1995) emphasize the strategic effect of exclusive distribution and show that two-part tariffs dominate wholesale price contracts.

Lin (1990) provides a comparison between independent and common retailing. He demonstrates that the competition-dampening strategic effect makes independent retailing more profitable for manufacturers. O'Brien and Shaffer (1993) point out that this is not the only reason why manufacturers prefer independent retailing: they show that a common retailer can keep part of the profits due to the threat of rejecting one contract offer. In contrast to all of these papers, our work considers a dynamic framework.

The comparison between distinct channel structures also motivates the literature on common agency. In their seminal paper, Bernheim and Whinston (1985) analyze an industry structure in which manufacturers delegate marketing activities to merchandise agents and offer different contracts for common and exclusive agents. They show that a common agent leads to collusive pricing and is therefore preferred by manufacturers. However, in contrast to Lin (1990) and O'Brien and Shaffer (1993), manufacturers have the outside option to hire another agent if their offer is rejected by a common agent.

Since then, several papers in the common agency literature offered explanations for why manufacturers may benefit from exclusive agents. Gal-Or (1991) assumes that agents have

¹³A detailed analysis of the implications of exclusive dealing is provided by Bernheim and Whinston (1998).

 $^{^{14}}$ Moorthy (1988) analyzes a generalization of the model and shows under which conditions this effect occurs.

private information about their costs that is correlated. She shows that a common agent can secure higher information rents due to the ability to provide manipulated information to both manufacturers and not just one. Besanko and Perry (1993) analyze the case in which manufacturers can make demand enhancing investments. In case of common agency, such investments entail positive spillovers on the rival. They show that this can induce manufacturers to choose exclusive dealing. Martimort (1996) considers a related model to Gal-Or (1991) in a more general setting. He confirms the finding that exclusive agency occurs if the information asymmetry between manufacturers and agents is strong (because the information rent kept by an exclusive agent is lower). Our paper provides another explanation for why manufacturers prefer exclusive distribution based on collusive considerations.

Finally, the paper also contributes to the literature on the observability of wholesale contracts. Coughlan and Wernerfelt (1989) and Caillaud and Rey (1995) show that the strategic effect, which dampens retail competition, is only present if wholesale contracts are observable.¹⁵ Manufacturers therefore prefer contracts that can be observed by the rival retailer.¹⁶ Our analysis demonstrates that this is no longer necessarily true in an infinitely repeated game given that the contractual form is a two-part tariff.

3 The Model

3.1 Players & Environment

Consider competition between two manufacturers, M_1 and M_2 , which sell imperfect substitute products. We distinguish between two channel structures: *i*) the goods are sold through a common retailer and *ii*) the goods are sold through independent retailers. The two structures are displayed in Figure 1. The retailing technology is one-to-one, and the final demand for manufacturer M_i 's product is $D^i(p_i, p_j)$, where p_i and p_j are the retail prices, with i, j = 1, 2 and $i \neq j$. The cost functions of manufacturers and retailers are linear with marginal cost normalized to zero.

 $^{^{15}}$ For a detailed analysis why secret wholesale contracts can nevertheless matter for market outcomes, see Katz (1991).

¹⁶Pagnozzi and Piccolo (2011) show that this result also hinges on the assumption on retailers' conjectures about their rivals' contracts. For example, the result can be reverted if retailers hold symmetric conjectures instead of the commonly used passive conjectures.





3.2 Contracts

Each manufacturer M_i offers a two-part tariff contract $C_i(w_i, T_i)$ either to his independent retailer R_i or to the common retailer R_c .¹⁷ This contract specifies the wholesale price w_i and the franchise fee T_i a retailer must pay to M_i when accepting the contract. A common retailer observes the offered contracts of both manufacturers before deciding about acceptance or rejection. If there are two independent retailers R_i and R_j , we consider both the scenario that a retailer can and cannot observe the contract offered to its rival.

3.3 Timing

We consider an infinitely repeated game with discrete time $t = 0, ..., \infty$. As in many papers in the literature on tacit collusion in vertically related markets (e.g., Jullien and Rey 2007, Nocke and White 2007, or Normann 2009), our focus is on manufacturer collusion. Manufacturers have a common discount factor $\delta \in [0, 1]$. In the main analysis, the retailer (or the retailers) maximizes spot profits.¹⁸ After the equilibrium analysis of both channel structures, we demonstrate that our insights remain valid if retailers are far-sighted as well.

The timing of events in the stage game is as follows. First, manufacturers simultaneously offer wholesale contracts $C_i(w_i, T_i)$ to either the common retailer R_c or the respective independent retailer R_i . Second, retailers accept or reject the contract offer(s). In case R_i rejects the contract offer, or R_c rejects both offers, the retailer obtains an outside option normalized to zero. Afterwards, retailers set downstream prices and the consumer market clears, that is, final demand materializes and input orders are placed.

¹⁷In what follows, we refer to a retailer by 'she' and to a manufacturer by 'he'.

¹⁸In fact, the competition policy guidelines on vertical market relationships both in the US and the EU focus exclusively on manufacturer collusion.

3.4 Tacit Collusion

If manufacturers collude, they maximize their discounted joint profits. We assume that collusion is sustained by Nash-reversion trigger strategies. That is, deviation by one manufacturer from collusion is punished by playing the Nash equilibrium of the stage game forever. We note that although manufacturers cannot observe contracts, a manufacturer can infer a deviation by the rival because he observes the input order of the retailer in each period.

We determine the value of the critical discount factor above which collusion is sustainable for each channel structure. We say that a structure facilitates collusion relative to the other if the critical discount factor is lower.

3.5 Assumptions & Equilibrium Concept

We develop our analysis under the following simplifying assumptions:

A1. The inverse demand function for product *i* is $P^i(q_i, q_j) = \alpha - \beta q_i - \gamma q_j$, for i = 1, 2, where q_i is product *i*'s total quantity.¹⁹ We assume that $\alpha > 0$ and $\beta > \gamma \ge 0$. The parameter γ measures the substitutability between products. The products are perfect substitutes if $\gamma = 1$, whereas each manufacturer is a monopolist if $\gamma = 0$. As there is price competition in the retail market, the system of inverse demand functions is inverted to get

$$D^{i}(p_{i}, p_{j}) = \frac{\alpha(\beta - \gamma) - \beta p_{i} + \gamma p_{j}}{\beta^{2} - \gamma^{2}} \quad \text{for} \quad i = 1, 2.$$

If only one product (i.e., product *i*) is sold, then $q_j = 0$ or $p_j \to \infty$, leading to an inverse demand function of $P^i(q_i, 0) = \alpha - \beta q_i$. Hence, the direct demand function is then $D^i(p_i, \infty) = (\alpha - p_i)/\beta$. This linear demand specification allows us to derive our results in the simplest possible way.

A2. A retailer will accept a contract offer when being indifferent between accepting and rejecting it. This assumption allows to restrict attention to equilibria with positive sales. Our solution concept is perfect Bayesian equilibrium because our model involves unobservable actions.

$$U(q_i, q_j) = \sum_{i=1}^{2} \left(\alpha q_i - \frac{1}{2} \beta q_i^2 \right) - \gamma q_1 q_2 - \sum_{i=1}^{2} p_i q_i + m,$$

where m denotes income. Maximizing the utility function with respect to q_i , i = 1, 2, yields the inverse demand function above.

 $^{^{19}\}mathrm{This}$ inverse demand results from a representative consumer with utility function

4 Common Retailer

Downstream Stage

We start with the case in which manufacturers distribute their products through a single retailer. When accepting the contract offers of both manufacturers, the common retailer's maximization program is given by

$$\pi_c(p_i, p_j) = D^i(p_i, p_j)(p_i - w_i) + D^j(p_j, p_i)(p_j - w_j) - T_i - T_j.$$
(1)

The first-order conditions providing a solution to (1) are given by the following system:²⁰

$$(p_i - w_i)\frac{\partial D^i(p_i, p_j)}{\partial p_i} + D^i(p_i, p_j) + (p_j - w_j)\frac{\partial D^j(p_j, p_i)}{\partial p_i} = 0, \quad i = 1, 2.$$
(2)

The optimal retail prices are not affected by the fixed fees.

The common retailer is not obliged to sell both products. She can reject one offer and distribute only the other product. Suppose R_c rejects the offer of M_i and sells only product j. Her profit function is then

$$\pi_c^j(w_j) = \max_{p_j} D^j(p_j, \infty)(p_j - w_j) - T_j.$$
 (3)

This implies that R_c optimally sets the monopoly price for product j given the wholesale price w_j . Let us denote this monopoly price by $p_j^M(w_j)$. R_c 's profit is then the monopoly profit minus the fixed fee T_j . Hence, although R_c does not have the power to offer contracts, threatening to reject a contract offer provides her with the opportunity to pit one manufacturer against the other, thereby allowing her to obtain a positive profit.

Upstream Stage and Collusion

In what follows, we first analyze the stage game equilibrium that arises along the punishment phase if manufacturer collusion breaks down. We then determine the collusive outcome.

In either case, R_c 's option to reject one offer implies that a representative manufacturer M_i can maximally extract his product's marginal contribution to R_c 's profit. This marginal contribution is the profit of R_c in case she accepts both offers, which is given by the optimized value of $\pi_c(p_i, p_j)$, minus the profit she obtains when only accepting M_j 's contract, which is given by (3). Denoting by $\mathbf{w} = (w_1, w_2)$ the vector of wholesale prices, M_i maximizes his

²⁰Second-order conditions are globally satisfied as long as D^i and D^j are concave or not strongly convex. Hence, they are satisfied with linear demand.

stage game profit

$$\Pi_i = D^i(p_i(\mathbf{w}), p_j(\mathbf{w}))w_i + T_i,$$

subject to the common retailer's participation constraint

$$D^{i}(p_{i}(\mathbf{w}), p_{j}(\mathbf{w}))(p_{i}(\mathbf{w}) - w_{i}) + D^{j}(p_{j}(\mathbf{w}), p_{i}(\mathbf{w}))(p_{j}(\mathbf{w}) - w_{j}) - T_{i} - T_{j} \ge \pi_{c}^{j}(w_{j}).$$
(4)

Evidently, the participation constraint must be binding in equilibrium. The optimization problem can then be written as

$$\max_{w_i} \ \Pi_i(\mathbf{w}) = D^i(p_i(\mathbf{w}), p_j(\mathbf{w})) p_i(\mathbf{w}) + D^j(p_j(\mathbf{w}), p_i(\mathbf{w})) (p_j(\mathbf{w}) - w_j) - T_j - \pi_c^j(w_j),$$

where the right-hand side of the equation is equivalent to

$$D^{i}(p_{i}(\mathbf{w}), p_{j}(\mathbf{w}))p_{i}(\mathbf{w}) + D^{j}(p_{j}(\mathbf{w}), p_{i}(\mathbf{w}))(p_{j}(\mathbf{w}) - w_{j}) - D^{j}(p_{j}^{M}(w_{j}), \infty)(p_{j}^{M}(w_{j}) - w_{j}).$$
 (5)

Given that both products are sold, the wholesale price w_c^N chosen by both manufacturers in the symmetric Nash equilibrium is derived from the following system of first-order conditions:

$$\left(\frac{\partial D^{i}(\cdot)}{\partial p_{i}} p_{i}(\mathbf{w}) + D^{i}(\cdot) + \frac{\partial D^{j}(\cdot)}{\partial p_{i}} (p_{j}(\mathbf{w}) - w_{j})\right) \frac{\partial p_{i}}{\partial w_{i}}$$

$$+ \left(\frac{\partial D^{i}(\cdot)}{\partial p_{j}} p_{i}(\mathbf{w}) + D^{j}(\cdot) + \frac{\partial D^{j}(\cdot)}{\partial p_{j}} (p_{j}(\mathbf{w}) - w_{j})\right) \frac{\partial p_{j}}{\partial w_{i}} = 0, \quad i = 1, 2,$$

$$(6)$$

where the arguments of $D^{i}(\cdot)$ and $D^{j}(\cdot)$ are those of the optimization program. Invoking the Envelope Theorem by using (2), we can rewrite (6) as

$$w_i\left(\frac{\partial D^i(\cdot)}{\partial p_i}\frac{\partial p_i}{\partial w_i} + \frac{\partial D^i(\cdot)}{\partial p_j}\frac{\partial p_j}{\partial w_i}\right) = 0, \quad i = 1, 2.$$

Since the bracket on the left-hand side of the last equation is negative, the equation can only be satisfied if manufacturers charge a wholesale price equal to marginal cost. Hence, the wholesale price in the symmetric Nash equilibrium is $w_c^N = 0$.

Each manufacturer sets a wholesale price as if he was integrated with the common retailer, maximizing the bilateral joint profit subject to the contract offered by the rival manufacturer (see O'Brien and Shaffer 1993). Because both manufacturers offer a wholesale price equal to marginal cost, there is no double marginalization. The common retailer therefore sets the same retail prices as a two-product monopolist. This implies that industry profits are maximized in the Nash equilibrium. The manufacturers and the common retailer share these profits.²¹ With linear demand, each manufacturer realizes an equilibrium profit of

$$\Pi_c^N = \frac{\alpha^2(\beta - \gamma)}{4\beta(\beta + \gamma)}.$$
(7)

It is evident that, if brands are perfect substitutes (i.e., $\gamma \rightarrow \beta$), manufacturers obtain zero profits. In the infinitely repeated game, the profit given by (7) is the profit along the punishment phase.

Consider now collusion between manufacturers. The retail prices for both products are always chosen by the common retailer according to (2) for given wholesale prices w_i and w_j . Even if manufacturers collude, they need to take into account R_c 's option to reject a contract. Therefore, the joint profit is $D^i(p_i(\mathbf{w}), p_j(\mathbf{w}))w_i + D^j(p_j(\mathbf{w}), p_i(\mathbf{w}))w_j + T_i + T_j$, where T_i and T_j are determined by the respective participation constraints. To determine the collusion profit, we can sum up the individual profits, given in (5), to get²²

$$\max_{(w_1,w_2)} D^i(p_i(\mathbf{w}), p_j(\mathbf{w})) p_i(\mathbf{w}) + D^j(p_j(\mathbf{w}), p_i(\mathbf{w})) p_j(\mathbf{w})$$
$$- \left[D^1(p_1^M(w_1), \infty) (p_1^M(w_1) - w_1) - D^1(p_1(\mathbf{w}), p_2(\mathbf{w})) (p_1(\mathbf{w}) - w_1) \right]$$
$$- \left[D^2(p_2^M(w_2), \infty) (p_2^M(w_2) - w_2) - D^2(p_2(\mathbf{w}), p_1(\mathbf{w})) (p_2(\mathbf{w}) - w_2) \right].$$

The first term is the industry profit, whereas the rent that must be left to the retailer is equal to the sum of the second and the third term. Using the Envelope Theorem, we obtain the system of first-order conditions that determines the solution to the above program:

$$\left(\frac{\partial D^{i}(\cdot)}{\partial p_{i}}p_{i}(\mathbf{w}) + D^{i}(\cdot) + \frac{\partial D^{j}(\cdot)}{\partial p_{i}}p_{j}(\mathbf{w})\right)\frac{\partial p_{i}}{\partial w_{i}} + \left(\frac{\partial D^{i}(\cdot)}{\partial p_{j}}p_{i}(\mathbf{w}) + D^{j}(\cdot) + \frac{\partial D^{j}(\cdot)}{\partial p_{j}}p_{j}(\mathbf{w})\right)\frac{\partial p_{j}}{\partial w_{i}} + D^{i}(p_{i}^{M}(w_{i}), \infty) - D^{i}(p_{i}(\mathbf{w}), p_{j}(\mathbf{w})) = 0, \quad i = 1, 2.$$
(8)

In the case of linear demand, the solution to (8) is

$$w_c^C = \frac{\alpha \gamma}{\beta + \gamma}$$

Because manufacturers are symmetric, they receive the same share of the aggregate collusion

 $^{^{21}}$ We provide a more detailed explanation how these shares change with the degree of competition below; see Lemma 1.

 $^{^{22}}$ As is standard in the analysis of collusion (see, e.g., Tirole 1988, or Nocke and White 2007), both manufacturers maximize joint profits by independently offering contracts to the retailer, and there is no contracting agency which proposes a single (bundled) contract to the retailer.

profit, which is

$$\Pi_c^C = \frac{\alpha^2 \beta}{4(\beta + \gamma)^2}.$$
(9)

In contrast to the Nash equilibrium, manufacturers set a wholesale price above marginal cost. This implies that manufacturers willingly accept double marginalization. Thus, industry profits are not maximized under collusion because the common retailer sets retail prices above the monopoly level.

To grasp the intuition behind the result, it is helpful to compare industry profits under collusion and in the stage game equilibrium and the respective shares obtained by manufacturers when varying the degree of competition:²³

Lemma 1. With a common retailer, the fiercer competition, the lower the industry profits under manufacturer collusion relative to those in the Nash equilibrium. However, as competition intensifies, the share of the industry profits obtained by manufacturers under collusion decreases to a smaller degree than in the Nash equilibrium.

The industry profits in the Nash equilibrium correspond to the maximized industry profits. This is independent of the level of competition because manufacturers set wholesale prices equal to zero and the common retailer acts as two-product monopolist. If competition gets fiercer (i.e., γ increases), it follows from (7) that the manufacturers' portion of the profits decreases and vanishes if products are perfect substitutes. The reason is that, at wholesale prices equal to marginal cost, an increase in competition makes the option of the common retailer to reject one manufacturer's contract and selling only its rival's product more valuable.

If the manufacturers collude, they aim to mitigate this effect, i.e., to lower the retailer's profit from pitting one manufacturer against another. They do so by raising the collusive wholesale price above marginal cost. This increase is larger the higher the degree of substitutability between the products. This strategy induces the retailer to increase downstream prices which causes industry profits to decrease. As a consequence, the industry profits under collusion are lower than those in the Nash equilibrium.

The lemma also shows that industry profits under collusion decrease to a larger extent than in the Nash equilibrium if competition becomes more intense. However, the share that manufacturers obtain is always higher under the former regime. This can be seen from (7) and (9), where, as competition intensifies, the decrease in the manufacturers' portion of the profits is larger in the Nash equilibrium than under collusion. For example, if products

 $^{^{23}}$ The proof of this Lemma and Lemma 2 in Section 7 is relegated to the Appendix.

become perfect substitutes (i.e., $\gamma \rightarrow 1$), this portion reaches a minimum of 2/3 under collusion but equals 0 in the stage game.

These results demonstrate that collusion with a common retailer works in a markedly different way as compared to a scenario in which firms sell directly to consumers. In the latter case, firms always seek to maximize industry profits.

The following proposition summarizes the preceding discussion.

Proposition 1. With a common retailer, each manufacturer realizes a profit from collusion that is given by (9). The industry profits under manufacturer collusion are lower than those under competition. However, manufacturers obtain a larger share.

Deviation and Sustainability of Collusion

To examine stability of the cooperative agreement between manufacturers, consider that one of them, say M_i , deviates from collusion. Suppose M_i offers a wholesale contract that departs from the collusive one, while M_j continues to offer the collusive contract. Assume first that deviation from the collusive agreement implies that the common retailer chooses to distribute only the deviant's product. The maximization program of the deviant is then given by

$$\max_{w_i} \ \Pi_i(w_i, w_c^C) = D^i(p_i^M(w_i), \infty) p_i^M(w_i) - \pi_c^j(w_c^C),$$
(10)

which yields the first-order condition

$$\frac{\partial p_i(\cdot)}{\partial w_i} \left(\frac{\partial D^i(\cdot)}{\partial p_i} p_i(\cdot) + D^i(\cdot) \right) = 0, \tag{11}$$

where the arguments of $D^i(\cdot)$ and $p_i(\cdot)$ are those given by (10). By the same arguments as above, the deviating manufacturer sets the wholesale price to maximize the joint bilateral profit. Therefore, the wholesale price solving (11) is given by $w_c^D = 0$, and the deviant obtains profits only through the franchise fee. The resulting deviation profit is

$$\Pi_c^D = \frac{\alpha^2 \left(\beta^2 (\beta + \gamma) + \gamma^2 (3\beta + \gamma)\right)}{4\beta (\beta + \gamma)^3}.$$
(12)

In Appendix A, we demonstrate that the optimal deviation in fact implies that only the deviant's product will be distributed in the retail market. This is intuitive: under the collusive regime, the retailer was indifferent between accepting or rejecting each contract. When deviating, a manufacturer offers a strictly lower wholesale price (i.e., $w_c^D < w_c^C$), which makes the non-deviating manufacturer's offer strictly less profitable for the common retailer. Equipped with this characterization, we can determine the critical discount factor above which manufacturers can sustain tacit collusion. The condition to determine this discount factor is standard, i.e., a manufacturer's profit stream from collusion must exceed the sum of profits realized in the deviation and the punishment phase. Formally,

$$\frac{\Pi_c^C}{1-\delta} \ge \Pi_c^D + \frac{\delta \Pi_c^N}{1-\delta}.$$

Using (7), (9) and (12) yields the following result:

Proposition 2. With a common retailer, manufacturers can sustain tacit collusion for all discount factors above

$$\delta_c = \frac{3\beta + \gamma}{2(2\beta + \gamma)}.\tag{13}$$

It is easy to show that the critical discount factor decreases in γ , i.e., collusion is easier to sustain when products are closer substitutes. The reason is (as is stated by Lemma 1) that the Nash profit falls to a larger degree than the collusion profit as competition becomes more intense, implying that the punishment becomes more severe. Conversely, δ_c increases in β , that is, cooperation is harder to sustain with a demand function that is less sensitive to own price changes.

We note that with a common retailer we do not need to distinguish between public or private contracts because the retailer observes the terms of both contracts and there is no competing retailer. By contrast, when turning to the case of independent retailers, the distinction between public and private contracts becomes important.

Far-Sighted Retailer

In our analysis, we assumed that the common retailer is short-lived, that is, her discount factor equals zero. We now demonstrate that our analysis carries over to the situation with a long-lived retailer (i.e., a retailer with discount factor $\delta \in (0, 1]$). The retailer can then also follow a grim-trigger strategy to obtain a higher share (or all) of the downstream profit. Such a grim-trigger strategy involves a punishment that the retailer inflicts upon manufacturers, after the manufacturers demanded too large a share of the profit. A punishment can either be carried out by rejecting one or both contract offers or by setting high retail prices, which lead to small quantities.²⁴

²⁴This setting follows the analysis of Gilo and Yehezkel (2016) who consider long-lived firms at both layers. In their model, retailers make offers and the grim-trigger strategy of the single manufacturer also consists of rejecting one or both contract offers.

However, the only scenario constituting a subgame perfect equilibrium at the retail level is the one in which the retailer sets prices according to (2) and accepts contracts adhering to the participation constraint (4). Since the collusion contracts meet this participation constraint and the retailer sets prices at the monopoly level, there is no opportunity for the retailer to punish manufacturers with a continuation play constituting a Nash equilibrium. As a consequence, even when being far-sighted, the retailer cannot do better but accepting the manufacturers' collusion contracts. Hence, the retailer has no credible grim-trigger strategy that leads to a breakdown of manufacturer collusion, implying that our results remains valid with a forward-looking retailer.

5 Independent Retailers

5.1 Private Contracts

Downstream Stage

Suppose now that manufacturers distribute their products through independent (exclusive) retailers. The retailers R_i and R_j independently set prices in the downstream market. We start with the case in which both retailers receive the contract offer from their respective manufacturer without observing the contract that is proposed to the rival. This implies that each retailer forms an expectation about the rival's retail price. To distinguish the notation with private contracts from the one with public contracts, we denote by p_j^e the expectation of R_i about the retail price of R_j . By contrast, in case of public offers we will just write p_j , as R_i observes w_j . The objective function of R_i with secret contracts is

$$\max_{p_i} D^i \left(p_i, p_j^e \right) \left(p_i - w_i \right) - T_i$$

Differentiating R_i 's objective function gives

$$(p_i - w_i) \frac{\partial D^i(p_i, p_j^e)}{\partial p_i} + D^i(p_i, p_j^e) = 0, \quad i = 1, 2.$$
(14)

The system (14) implicitly determines the optimal retail prices denoted by $p_i(w_i, p_i^e)$.

Upstream Stage and Collusion

Using backward induction, we can now solve the upstream game. As in the previous section, we start with the analysis of the stage game that yields the manufacturers' profits along the punishment phase. Manufacturer M_i offers a contract to R_i that maximizes his

profit, subject to R_i 's participation constraint. The participation constraint is given by

$$D^{i}\left(p_{i}(w_{i}, p_{j}^{e}), p_{j}^{e}\right)\left(p_{i}(w_{i}, p_{j}^{e}) - w_{i}\right) - T_{i} \geq 0.$$

In contrast to the common retailer, R_i does not have the option to reject M_i 's offer and sell a different product. Therefore, M_i can extract R_i 's entire profit via the franchise fee. The maximization program of M_i is

$$\max_{w_i} \ \Pi_i(w_i) = D^i\left(p_i(w_i, p_j^e), p_j^e\right) p_i(w_i, p_j^e).$$

Differentiating M_i 's objective function gives

$$\frac{\partial p_i(w_i, p_j^e)}{\partial w_i} \left(\frac{\partial D^i\left(p_i(w_i, p_j^e), p_j^e\right)}{\partial p_i} p_i(w_i, p_j^e) + D^i\left(p_i(w_i, p_j^e), p_j^e\right) \right) = 0$$

In a symmetric equilibrium, both manufacturers offer the same contract and expectations are fulfilled. Denoting the equilibrium wholesale price by w_I^N , this implies $p_j^e = p_j(w_I^N)$. Inserting this into the last equation, we obtain that w_I^N satisfies

$$\frac{\partial p_i(w_I^N)}{\partial w_i} \left(\frac{\partial D^i\left(p_i(w_I^N), p_j(w_I^N)\right)}{\partial p_i} p_i(w_I^N) + D^i\left(p_i(w_I^N), p_j(w_I^N)\right) \right) = 0$$

Using (14), it is straightforward to show that the optimal wholesale price is equal to marginal cost, that is, $w_I^N = 0$. Since a retailer cannot observe the contract proposed by the rival manufacturer, each manufacturer acts as if integrated with his retailer. With linear demand, the profit along the punishment phase realized by each manufacturer is given by

$$\Pi_I^N = \frac{\alpha^2 \beta (\beta - \gamma)}{(\beta + \gamma)(2\beta - \gamma)^2}.$$
(15)

In contrast to the common retailer, a wholesale price of zero does not maximize industry profits with independent retailers. Independent retailers are competitors at the downstream level. Consequently, they are not able to set monopoly retail prices if manufacturers charge them a wholesale price equal to marginal cost.

Consider now collusion. Retailers set prices according to (14) for given wholesale prices. Manufacturers anticipate the retailers' reactions and their objective function is

$$\max_{(w_1,w_2)} D^i(p_i(w_i), p_j(w_j)) p_i(w_i) + D^j(p_j(w_j), p_i(w_i)) p_j(w_j).$$

The wholesale prices solving the manufacturers' maximization problem are determined by the following first-order conditions:

$$\frac{\partial p_i(w_i)}{\partial w_i} \left(\frac{\partial D^i\left(p_i(w_i), p_j(w_j)\right)}{\partial p_i} p_i(w_i) + D^i\left(p_i(w_i), p_j(w_j)\right) + \frac{\partial D^j\left(p_j(w_j), p_i(w_i)\right)}{\partial p_i} p_j(w_j) \right) = 0.$$

With linear demand, the symmetric collusive wholesale price is

$$w_I^C = \frac{\alpha \gamma}{2\beta}.\tag{16}$$

Colluding manufacturers choose the wholesale price to induce retailers to set the final price at the monopoly level $(p_I^M = \alpha/2)$ independent of the degree of differentiation. Each manufacturer's profit from collusion is

$$\Pi_I^C = \frac{\alpha^2}{4(\beta + \gamma)}.\tag{17}$$

When they tacitly collude with independent retailers, manufacturers, in fact, choose a strategy that aims at maximizing industry profits. Since the retailers have no outside option, the only strategic purpose of colluding manufacturers is to induce retailers to set the monopoly retail prices. Manufacturers can do so by raising their wholesale prices above marginal cost. Thus, the competing retailers face higher per-unit costs and optimally increase the retail price. Manufacturers obtain the entire industry profits via the fixed fee and the wholesale margin. Therefore, tacit collusion between manufacturers distributing through independent retailers works in similar way as tacit collusion between firms that directly sell to consumers: industry profits are maximized under collusion. The results are highlighted in the following proposition.

Proposition 3. With independent (exclusive) retailers, each manufacturer realizes a profit from collusion that is given by (17). The collusion profits correspond to the maximized industry profits, which exceed those under competition.

Deviation and Sustainability of Collusion

Applying the same logic developed above, we can determine the deviation wholesale price and profit. The objective function of the deviant is

$$\max_{w_i} \ \Pi_i(w_i, w_I^C) = D^i(p_i(w_i), p_j(w_I^C)) p_i(w_i),$$

leading to a first-order condition of

$$\frac{\partial p_i(w_i)}{\partial w_i} \left(\frac{\partial D^i\left(p_i(w_i), p_j(w_I^C)\right)}{\partial p_i} p_i(w_i) + D^i\left(p_i(w_i), p_j(w_I^C)\right) \right) = 0$$

Again, using (14), it is easy to see that the deviant's optimal wholesale price is $w_I^D = 0$. Because contracts are unobservable, M_i 's contract choice has no impact on the price of retailer *j*. This implies that M_i sets the wholesale price to maximize channel profits. A wholesale price of zero avoids double marginalization and, therefore, aligns the incentives of the manufacturer and the retailer. The resulting profit realized by the deviant is given by

$$\Pi_I^D = \frac{\alpha^2 (2\beta - \gamma)^2}{16\beta(\beta - \gamma)(\beta + \gamma)}.$$
(18)

Before we can determine the critical discount factor above which collusion is sustainable, we need to check whether a deviation leads to zero demand for the product of the nondeviating manufacturer. If M_i deviates, he offers w_I^D to R_i , while M_j sticks to w_I^C . The resulting demand for product j in the retail market is

$$D^{j}(w_{I}^{C}, w_{I}^{D}) = \frac{\alpha \left(2\beta(\beta - \gamma) - \gamma^{2}\right)}{4\beta(\beta + \gamma)(\beta - \gamma)}$$

Setting $D^j(w_I^C, w_I^D)$ equal to zero shows that the demand for product j is positive if and only if $\gamma \leq \hat{\gamma} \equiv (\sqrt{3} - 1)\beta \approx (0.732)\beta^{25}$ Thus, the analysis of independent retailers conducted so far only applies if the products are sufficiently differentiated.

For $\gamma > \hat{\gamma}$, the retailer of the deviating manufacturer M_i optimally sets her retail price such that the demand for product j equals zero. With $p_j = p_I^M = \alpha/2$, this is achieved with a retail price of $p_i = \hat{p}_I^D = (\alpha(2\gamma - \beta))/(2\gamma)$. The corresponding deviation profit realized by M_i is²⁶

$$\widehat{\Pi}_{I}^{D} = \frac{\alpha^{2}(2\gamma - \beta)}{4\gamma^{2}}.$$
(19)

We can now determine the critical discount factor by inserting (15), (17), (18) and (19) into the condition $\Pi_I^C/(1-\delta) \ge \Pi_I^D + \delta \Pi_I^N/(1-\delta)$.

²⁵Here and in what follows, we round the thresholds to three decimal places to simplify the exposition. ²⁶It can be easily verified that $\Pi_I^D = \widehat{\Pi}_I^D$ for $\gamma = (\sqrt{3} - 1)\beta$.

Proposition 4. With independent (exclusive) retailers and unobservable contracts, manufacturers can sustain tacit collusion for all discount factors above

$$\delta_{I} = \begin{cases} \frac{(2\beta-\gamma)^{2}}{8\beta(\beta-\gamma)+\gamma^{2}} & \text{for } \gamma \in [0,\widehat{\gamma}] \\ \frac{(2\beta-\gamma)^{2} (\beta(\beta-\gamma)-\gamma^{2})}{4\beta^{3}(\beta-2\gamma)+\gamma^{2}(\beta(\beta+3\gamma)-2\gamma^{2})} & \text{for } \gamma \in (\widehat{\gamma},\beta). \end{cases}$$
(20)

In contrast to the critical discount factor with a common retailer, (20) is an increasing function in γ if $\gamma \leq \hat{\gamma}$, that is, collusion becomes more difficult to sustain when products are closer substitutes. Conversely, it is a decreasing function in β , that is, collusion is facilitated if the impact of own price variations on the demand is low. If $\gamma > \hat{\gamma}$, the critical discount factor changes non-monotonically with γ and β . It increases in γ if $\gamma \leq (0.745)\beta$ and decreases if $\gamma > (0.745)\beta$, whereas the opposite slope occurs for β . Therefore, if products are close substitutes, δ_I is affected by γ and β in the same way as δ_c , whereas the opposite holds true for moderate or low degrees of competition.

Far-Sighted Retailers

As in the previous section, we derived our results under the assumption that retailers maximize only spot profits. Consider now that retailers are far-sighted and discount future profits with a common discount factor. In case they collude, they *jointly* set retail prices and decide on accepting or rejecting contracts. In particular, under the assumption that retailers can make side-payments, they will coordinate their acceptance decisions so that joint profits are maximized and that the split of profits minimizes deviation incentives.²⁷ In case that the discount factor is sufficiently large, the collusive strategy of the retailers is therefore identical to the one of the common retailer. This implies that each manufacturer, when dealing with his retailer, needs to take participation constraint (4) into account. In contrast, if the critical discount factor is small, the retail cartel will break down since retailers are not patient enough to sustain collusion. Thus, depending on the discount factor with which retailers discount future profits, either our analysis of the common retailer or of short-lived independent retailers applies.²⁸

²⁷Side payments between colluding firms are well established in the literature (see Tirole 1988 or Athey and Bagwell 2001, among many others).

²⁸Note that for intermediate values of the retailers' discount factor, manufacturers may set wholesale prices to just destroy the retail cartel. Even if this implies wholesale prices slightly different from those derived above, the qualitative results remain valid.

5.2 Public Contracts

Downstream Stage

Consider now that retailers observe both contract offers before entering competition. This may be the case, for instance, if disclosure standards are mandatory or if manufacturers can easily form information-sharing agreements. In addition, supplier trade associations may facilitate transmission of information among competing distribution channels. For example, information-intensive channels, where channel members invested in information technologies such as telecommunication and satellite linkages or bar coding, became more important in recent years (see e.g., Palmatier et al. 2016).

Under public contracts, retailer R_i chooses her retail price to maximize $D^i(p_i, p_j)(p_i - w_i) - T_i$. The second-stage equilibrium is determined by the first-order conditions

$$\frac{\partial D^{i}(p_{i}, p_{j})}{\partial p_{i}}(p_{i} - w_{i}) + D^{i}(p_{i}, p_{j}) = 0, \quad i = 1, 2.$$
(21)

This system can be solved for the retailers' best response functions, i.e., $p_i(w_i, w_j)$, i = 1, 2, which are increasing in both w_i and w_j . In contrast to private contracts, a retailer's optimal retail price depends on both wholesale prices because each retailer observes the contract proposed to the rival retailer.

Upstream Stage and Collusion

In the upstream game, each manufacturer chooses the franchise fee such that the participation constraint of his retailer is met. Hence, M_i sets his wholesale price to maximize

$$\max_{w_i} \ \Pi_i(w_i, w_j) = D^i(p_i(w_i, w_j), p_j(w_j, w_i)) p_i(w_i, w_j).$$
(22)

The wholesale price in the symmetric Nash-equilibrium of the upstream game is therefore implicitly defined by the following first-order conditions:

$$\left(\frac{\partial D^{i}\left(\cdot\right)}{\partial p_{i}} p_{i}\left(\cdot\right) + D^{i}\left(\cdot\right)\right) \frac{\partial p_{i}\left(\cdot\right)}{\partial w_{i}} + \frac{\partial D^{i}\left(\cdot\right)}{\partial p_{j}} \frac{\partial p_{j}\left(\cdot\right)}{\partial w_{i}} p_{i}\left(\cdot\right) = 0, \quad i = 1, 2,$$
(23)

where the arguments of $D^i(\cdot)$, $p_i(\cdot)$, and $p_j(\cdot)$ are those of (22). In contrast to private contracts, M_i now takes R_j 's reaction into account when setting the wholesale price. This strategic effect is represented by the second term of (23), which is positive because prices are strategic complements. Hence, raising w_i leads to an increase in p_j which positively affects M_i 's profit. Consequently, the wholesale prices chosen by manufacturers in the stage game along the punishment phase are above marginal cost. With linear demand, the equilibrium wholesale price with independent retailing and observable contracts is given by

$$w_{IO}^{N} = \frac{\alpha \gamma^{2} (\beta - \gamma)}{\beta \left(4\beta^{2} - \gamma^{2} - 2\beta\gamma\right)}.$$

The resulting manufacturer profit is given by

$$\Pi_{IO}^{N} = \frac{2\alpha^{2}\beta(\beta-\gamma)\left(2\beta^{2}-\gamma^{2}\right)}{\left(\beta+\gamma\right)\left(4\beta^{2}-\gamma^{2}-2\beta\gamma\right)^{2}}.$$

If $\gamma \rightarrow \beta$, both manufacturers price at marginal cost and realize zero profits.

Consider now that manufacturers collude. The retailers' optimal responses to the wholesale prices that maximize the manufacturers' joint profits are still determined by (21). As under private contracts, it can be easily shown that manufacturers offer contracts to their retailers that induce them to set the monopoly retail price. Hence, with linear demand, the optimal collusive wholesale price w_{IO}^C is identical to (16), and the resulting profit Π_{IO}^C that each manufacturer receives in equilibrium is likewise given by (17). Collusion under observable contracts works in a similar way as under private contracts: manufacturers seek to maximize industry profits which they fully extract from their retailers. However, strategies to punish a deviation differ. Due to the strategic effect, manufacturers realize higher stage game profits if contract offers are observable to the rival retailer.

Deviation and Sustainability of Collusion

When deviating from the collusive agreement, M_i faces a maximization program given by

$$\max_{w_i} \ \Pi_i(w_i, w_{IO}^C) = D^i\left(p_i(w_i, w_{IO}^C), p_j(w_{IO}^C, w_i)\right) p_i(w_i, w_{IO}^C).$$
(24)

The first-order condition is determined by (23) where the arguments of $D^{i}(\cdot)$, $p_{i}(\cdot)$ and $p_{j}(\cdot)$ are replaced by those of (24). With linear demand, the optimal deviation wholesale price is

$$w_{IO}^{D} = \frac{\alpha \gamma^2 \left(4\beta^2 - 2\beta\gamma - \gamma^2\right)}{8\beta^2 \left(2\beta^2 - \gamma^2\right)}.$$

In contrast to the case with private contracts, R_j now immediately responds to M_i 's deviation by lowering the retail price. This induces M_i to undercut the collusive wholesale price to a smaller extent than in case of private contracts. Under observable contracts, the deviation wholesale price exceeds marginal cost and the resulting deviation profit is given by

$$\Pi_{IO}^{D} = \frac{\alpha^2 \left(4\beta^2 - 2\beta\gamma - \gamma^2\right)^2}{32\beta \left(\beta + \gamma\right) \left(\beta - \gamma\right) \left(2\beta^2 - \gamma^2\right)}.$$

Proceeding in the same way as above, we check if deviation of M_i implies that M_j 's demand becomes zero when competition gets fierce. We obtain that this occurs for $\gamma > (0.841)\beta \equiv \tilde{\gamma}$. For $\gamma > \tilde{\gamma}$, M_i optimally sets a deviation wholesale price given by

$$\widetilde{w}_{IO}^{D} = \frac{\alpha \left(2\beta \left(2\beta (\gamma - \beta) + \gamma^{2}\right) - \gamma^{3}\right)}{2\beta^{2} \gamma},$$

which results in a deviation profit of

$$\widetilde{\Pi}_{IO}^{D} = \frac{\alpha^{2} \left(4\beta \left(2\gamma - \beta\right) - 3\gamma^{2}\right)}{4\beta\gamma^{4}}.$$

Now we have all relevant information to determine the critical discount factor above which collusion is sustainable if contracts are observable. The solution to the standard indifference condition is provided in the next proposition.

Proposition 5. With independent (exclusive) retailers and under observable contracts, manufacturers can sustain tacit collusion for all discount factors above

$$\delta_{IO} = \begin{cases} \frac{\left(2\beta(2\beta-\gamma)-\gamma^2\right)^2}{4\beta(\beta-\gamma)(8\beta^2-3\gamma^2)+\gamma^4} & \text{for } \gamma \in [0,\widetilde{\gamma}] \\ \frac{\left(2\beta(2\beta-\gamma)-\gamma^2\right)^2\left(3\gamma^3+4\beta\left(\beta(\beta-\gamma)-\gamma^2\right)\right)}{3\gamma^7+\beta(7\gamma^6+4\beta(4\beta(\beta(9\gamma^3+\beta(4\beta^2-8\beta\gamma-\gamma^2))-3\gamma^4)-7\gamma^5)))} & \text{for } \gamma \in (\widetilde{\gamma},\beta). \end{cases}$$
(25)

6 Stability of Tacit Collusion and Welfare Implications

6.1 Stability

Following the analysis of Sections 4 and 5, the natural question arises which channel structure is most suitable for manufacturers to sustain tacit collusion. Using the critical discount factors obtained in *Proposition 2*, *Proposition 4* and *Proposition 5*, we start with the comparison between common retailing and independent retailing under both observable and private contracts.²⁹

Proposition 6. With linear demand, each manufacturer realizes a higher collusion profit with independent retailers than with a common retailer. In addition, the critical discount factor above which collusion can be sustained is lower when distributing through

²⁹The proofs of all subsequent propositions are relegated to the Appendix.

independent retailers as compared to selling through a common retailer. This result holds regardless of whether contracts are private or observable with independent retailers. Formally, $\delta_I < \delta_c$ and $\delta_{IO} < \delta_c$.

Distribution through independent retailers allows manufacturers to sustain collusion for a larger range of discount factors irrespective of whether contracts are observable or not. To develop the intuition behind this result, we proceed by analyzing the difference between the discount factors with general demand functions for the case in which brands are close substitutes.

We start with independent retailers and focus on private contracts.³⁰ Since manufacturers set their wholesale prices equal to marginal cost in the Nash equilibrium, the profit along the punishment phase is given by $\Pi_I^N = D^i(p_i(0), p_j(0))p_i(0)$. By contrast, in the collusion phase, each manufacturer obtains the monopoly profit, which is $\Pi_I^C = D^i(p_i^M, p_j^M)p_i^M$. Finally, the deviation profit of M_i is $\Pi_I^D = D^i(p_i(0), p_j^M)p_i(0)$. If products become close substitutes, $\Pi_I^N \to 0$ as $p_i(0) \to 0$. When colluding, each manufacturer obtains half of the maximized industry profits, while, under high product substitutability, Π_I^D approaches the total maximized industry profits, as the deviant marginally undercuts the collusive downstream price so that the rival product is driven out of the market. Inserting this in the formula for the critical discount factor yields the well-known result that $\delta_I \to 1/2$ due to the fact that $(\Pi_I^D - \Pi_I^C)/(\Pi_I^D - \Pi_I^N) \to 1/2$.

We now turn to the common retailer. Along the punishment phase, each manufacturer sets $w_c^N = 0$. Using (5) and symmetry of manufacturers, the Nash profit of a representative manufacturer M_i can be written as $\Pi_c^N = 2D^i(p_i(0,0), p_j(0,0))p_i(0,0) - D^j(p_j^M(0), \infty)p_j^M(0)$. Under collusion, M_i gets $D^i(p_i(w_c^C, w_c^C), p_j(w_c^C, w_c^C))w_c^C + T_i^C$, where T_i^C equals

$$2D^{i}\left(p_{i}(w_{c}^{C},w_{c}^{C}),p_{j}(w_{c}^{C},w_{c}^{C})\right)\left(p_{i}(w_{c}^{C},w_{c}^{C})-w_{c}^{C}\right)-D^{j}\left(p_{j}^{M}(w_{c}^{C}),\infty\right)\left(p_{j}^{M}(w_{c}^{C})-w_{c}^{C}\right)$$

When M_i deviates from this collusive agreement, he sets $w_c^D = 0$ and the common retailer only accepts the deviant's offer. The deviant's profit is therefore equal to the maximized industry profit minus the profit that the common retailer obtains in case she accepts only the offer of the manufacturer who sticks to collusion. The latter is $D^j \left(p_j^M(w_c^C), \infty \right) \left(p_j^M(w_c^C) - w_c^C \right) - T_j^C$. Thus, Π_c^D is

$$D^{i}(p_{i}(0), \infty)p_{i}(0) + 2D^{i}(p_{i}(\cdot), p_{i}(\cdot))\left(p_{j}(\cdot) - w_{c}^{C}\right) - 2D^{j}\left(p_{j}^{M}(w_{c}^{C})\right)\left(p_{j}^{M}(w_{c}^{C}) - w_{c}^{C}\right) + 2D^{i}\left(p_{i}(\cdot), p_{i}(\cdot)\right)\left(p_{j}(\cdot) - w_{c}^{C}\right) - 2D^{j}\left(p_{j}^{M}(w_{c}^{C})\right)\left(p_{j}^{M}(w_{c}^{C}) - w_{c}^{C}\right) + 2D^{i}\left(p_{i}(\cdot), p_{i}(\cdot)\right)\left(p_{j}(\cdot) - w_{c}^{C}\right) - 2D^{j}\left(p_{j}^{M}(w_{c}^{C})\right)\left(p_{j}^{M}(w_{c}^{C}) - w_{c}^{C}\right) + 2D^{i}\left(p_{j}(\cdot), p_{i}(\cdot)\right)\left(p_{j}(\cdot) - w_{c}^{C}\right) - 2D^{j}\left(p_{j}^{M}(w_{c}^{C})\right)\left(p_{j}^{M}(w_{c}^{C}) - w_{c}^{C}\right) + 2D^{i}\left(p_{j}(\cdot), p_{i}(\cdot)\right)\left(p_{j}(\cdot) - w_{c}^{C}\right) + 2D^{i}\left(p_{j}(\cdot), p_{i}(\cdot)\right)\left(p_{j}(\cdot) - w_{c}^{C}\right) + 2D^{i}\left(p_{j}(\cdot), p_{i}(\cdot)\right)\left(p_{j}(\cdot) - w_{c}^{C}\right) + 2D^{i}\left(p_{j}(\cdot), p_{i}(\cdot)\right)\left(p_{j}(\cdot), p_{i}(\cdot)\right)\left(p_{j}(\cdot), p_{i}(\cdot)\right)\left(p_{j}(\cdot), p_{i}(\cdot)\right)\right)$$

where $(\cdot) = (w_c^C, w_c^C)$.

³⁰The analysis with public contracts follows very similar lines.

If brands become close substitutes, $D^i(p_i(0,0), p_j(0,0))p_i(0,0)$ converges to $D^j(p_j^M(0), \infty)$ $p_j^M(0)/2$ (i.e., half of the monopoly profit), as the common retailer sets the monopoly price for each product. This implies that $\Pi_c^N \to 0$. Therefore, the critical discount factor under common retailing approaches $\delta_c = 1 - \Pi_c^C / \Pi_c^D$.

We can now determine whether δ_c is larger than 1/2, which is equivalent to $\Pi_c^C/\Pi_c^D < 1/2$. Inserting the respective expressions for the profits and rearranging yields that this holds if

$$D^{i}(p_{i}(0), \infty)p_{i}(0) > 2D^{i}\left(p_{i}(w_{c}^{C}, w_{c}^{C}), p_{j}(w_{c}^{C}, w_{c}^{C})\right)p_{i}(w_{c}^{C}, w_{c}^{C}).$$
(26)

At high product substitutability, the left-hand side is equal to the maximized industry profits as the wholesale price approaches marginal cost, thereby avoiding double marginalization. By contrast, the term on the right-hand side involves industry profits with wholesale prices above marginal cost (i.e., at $w_c^C > 0$). Given that product are close substitutes, the righthand side is therefore close to $D^i(p_i^M(w_c^C), \infty) p_i^M(w_c^C)$, which must consequently be lower than $D^i(p_i(0), \infty)p_i(0)$.

The above discussion demonstrates that the ratio between deviation and collusion profits is larger under common retailing than under independent retailing. The intuition behind this result lies in the common retailer's option to reject M_i 's contract and accept only the contract of the rival manufacturer. The value of this option is the same under collusion and under deviation because the competing manufacturer sticks to collusion and offers the same contract independent of M_i 's action. When deviating, M_i obtains the approximately maximized industry profits minus the retailer's outside option because the retailer optimally accepts only the deviant's offer. In contrast, when sticking to collusion, M_i obtains *less* than half of the maximized industry profits minus the retailer's outside option. This is due to the fact that manufacturers collude by raising their wholesale prices to reduce the retailer's threat of pitting them against each other.³¹ Instead, with independent retailers, each manufacturer obtains approximately maximized industry profits when deviating and half of them when colluding. As a consequence, deviation incentives are higher with a common retailer than with independent retailers.

We derived the intuition for the case in which brands are close substitutes. With strictly differentiated products, the punishment profit is no longer approximately zero, which makes the comparison between the two channel structures with general demand complex. However, the main intuition for our result carries over to this case. As we have shown with linear demand, the collusion profit with common retailing is also below the maximum in this case whereas deviation leads to monopolization of the downstream market. Although the

³¹This is reflected by the right-hand side of (26), which involves wholesale prices of w_c^C , whereas the wholesale price on the left-hand side equals 0.

punishment is then less severe under independent retailing than under common retailing, the effect of the stronger deviation incentives with a common retailer dominates, implying that independent retailing facilitates collusion.

We now examine whether observability of contracts facilitates collusion for manufacturers in case of independent retailing. The following proposition shows that this is not necessarily true.

Proposition 7. If manufacturers distribute through independent retailers, public contracts make manufacturers' collusion harder to sustain compared to private contracts if and only if competition is fierce, i.e., if and only if $\gamma > 0.825\beta$.

With independent retailers, contract observability facilitates collusion if competition is not particularly strong and impedes it otherwise. The intuition for this result arises from the strategic effect and its implications for the deviation and punishment profit. If a manufacturer deviates from the collusive agreement and contracts are observable, the rival retailer detects the deviation in the very same period and can therefore react immediately. Because downstream prices are strategic complements, the rival retailer's optimal reaction is to lower the downstream price, which renders deviation less profitable. In contrast, under private contracts, the deviation cannot be observed in the same period and the rival retailer continues to set the collusive downstream price. Consequently, the deviation profit is higher under private contracts. This effect makes the cartel more stable under public contracts.

However, the opposite applies to the punishment phase where the strategic effect dampens competition in the Nash equilibrium. Under observable contracts, a manufacturer can induce the competing retailer to increase the downstream price by proposing a higher wholesale price to his own retailer. That is, a higher w_i results in a higher downstream price of R_i and, due to the strategic complementary of the downstream prices, R_j reacts by increasing her downstream price as well. Consequently, Nash profits are higher under public contracts and punishment is less severe. This effect works in favor of collusion under private contracts.

As competition gets fiercer (i.e., γ increases), the first effect loses significance relative to the second effect. The retailer of the non-deviating manufacturer is constrained in her reaction by the relatively high collusive wholesale price. Although being able to react in the period of deviation, the retailer is a weak competitor, and if products are close enough substitutes, she can no longer obtain positive sales. Consequently, the effect of the immediate reaction to deviation becomes less important relative to the difference in punishment profits. Hence, private contracts facilitate collusion if competition is fierce. By contrast, for low or intermediate levels of competition, contract observability is more suitable for collusive purposes because the ability to detect and to react to a deviation dominates.

Table 1 summarizes the manufacturers' equilibrium profits for each channel structure, and the resulting critical discount factors above which they can sustain collusion. To facilitate the comparison, β is set equal to 1.

Table 1:	Equilibrium	$\operatorname{profits}$	and	critical	discount	factors	for	different	levels	of
			con	npetition	n ($\beta = 1$)					

	Punishment profit (Π^N)	Collusion profit (Π^C)	Deviation profit (Π^D)	Critical discount factor
Common retailer	$\frac{\alpha^2(1-\gamma)}{4(1+\gamma)}$	$\frac{\alpha^2}{4(1+\gamma)^2}$	$\frac{\alpha^2(1+\gamma+\gamma^2(3+\gamma))}{4(1+\gamma)^3}$	$\begin{array}{ll} 0.7 & (\gamma=0.5) \\ 0.672 & (\gamma=0.9) \end{array}$
Independent retailers — secret offers	$\frac{\alpha^2(1-\gamma)}{(1+\gamma)(2-\gamma)^2}$	$\frac{\alpha^2}{4(1+\gamma)}$	$\frac{\frac{\alpha^2 (2-\gamma)^2}{16(1-\gamma^2)}}{\frac{\alpha^2 (2\gamma-1)}{4\gamma^2}} (\gamma \le 0.732)$	0.529 ($\gamma = 0.5$) 0.567 ($\gamma = 0.9$)
Independent retailers — public offers	$\frac{2\alpha^2(1-\gamma)\left(2-\gamma^2\right)}{(1+\gamma)\left(4-\gamma^2-2\gamma\right)^2}$	$\frac{\alpha^2}{4(1+\gamma)}$	$\frac{\frac{\alpha^2(4-2\gamma-\gamma^2)^2}{32(2-\gamma^2(3-\gamma^2))}}{\frac{\alpha^2(4(2\gamma-1)-3\gamma^2)}{4\gamma^4}} (\gamma > 0.841)$	0.519 ($\gamma = 0.5$) 0.614 ($\gamma = 0.9$)

Interestingly, our results reveal that, whereas public contracts are always preferred from the perspective of static (or one-shot) competition, this is no longer true when dynamic considerations come into play. In this respect, our model provides a strategic explanation for why manufacturers want to keep their contracts secret. There may be several reasons for not disclosing wholesale contracts voluntarily, e.g., privacy concerns on input costs or production flows. However, from a strategic perspective, such secrecy is hard to justify. Our model shows that disclosing wholesale contracts can lead to a breakdown of collusion (due to reduced punishment threats) if products are close enough substitutes. Figure 2 plots the discount factors above which manufacturers can form a stable tacit cartel for the respective channel structures.

We also note that if retailers were to make contract offers and collusion takes place only at the retail level, observable contracts would still be unambiguously preferred over secret ones by retailers. This result is obtained in Piccolo and Miklòs-Thal (2012). The intuition is twofold: first, with secrecy each retailer acts as if vertically integrated with its manufacturer since no strategic effect is present. This implies that manufacturers do not serve as a collusion-facilitating device for retailers. Second, similar to our paper, observable contracts allow for an reaction to a deviation in the very same period, whereas this is not possible with secret contracts. However, in contrast to our paper, the punishment profit is the same with observability and secrecy because retailers always offer a contract involving marginal-cost pricing in the Nash equilibrium. Instead, we show that in case manufacturers make contract offers, punishment is less severe with public contracting, which can facilitate collusion under secret contracting.





6.2 Welfare

This section discusses the welfare implications of the findings. From the result of *Proposition 6*, one may suggest that independent retailing has a larger detrimental effect on social welfare than common retailing and should therefore be subject to scrutiny by the antitrust authorities. However, this neglects that the common retailer operates as a downstream monopolist.

Consider first the case in which collusion is not sustainable in any regime, that is, $\delta < \delta_I < \delta_c$. In this scenario, the common retailer gets the inputs at marginal cost and acts as two-product monopolist in the consumer market. This is not the case with independent retailers. Although they also buy the input from their manufacturer at marginal cost, they face competition from their rival, which drives retail prices below the monopoly level. In fact, $p_c^N = \alpha/2 > \alpha(\beta - \gamma)/(2\beta - \gamma) = p_I^N$. Second, if $\delta_I < \delta_c \leq \delta$ collusion is sustainable under either regime. It is straightforward that common retailing is again more harmful than independent retailing in this scenario. The reason is that colluding manufacturers optimally raise the wholesale price to diminish the retailer's opportunity to reject one contract. This collusive strategy involves double marginalization and induces the common retailer to increase prices above the monopoly level. We obtain $p_c^C = \alpha(\beta + 2\gamma)/(\beta + \gamma) > \alpha/2 = p_I^C$. Finally, in case that $\delta_I \leq \delta < \delta_c$, i.e., collusion is sustainable with independent retailers but not with a common retailer, final good prices are the same under both regimes, that is, $p_c^N = \alpha/2 = p_I^C$. Under common retailing, wholesale prices are equal to marginal cost and the retailer sets monopoly prices for the two brands. Similarly, under independent retailing, colluding manufacturers set wholesale prices in such a way that their retailers charge monopoly prices.

The latter result is due to the fact that we focused on full collusion. Allowing for partial collusion again yields the result that common retailing is more detrimental to welfare. To see this, note that for $\delta_I \leq \delta < \delta_c$, manufacturers achieve full collusion with independent retailing; hence, considering partial collusion does not change the result. By contrast, under common retailing partial collusion allows manufacturers to raise wholesale prices above marginal cost, which again leads to retail prices above the monopoly level.

As a consequence, welfare is always higher under independent retailing, and this result holds although collusion with independent retailers can be sustained for a larger range of discount factors. The effect that a common retailer monopolizes the downstream market is dominant.

6.3 Outside Option for Retailers

In our analysis, we assumed that retailers do not have an outside option, that is, a retailer can only distribute product i when accepting the offer of manufacturer i but there is no other firm offering the product. In this subsection, we briefly demonstrate that our main results are robust to the introduction of an outside option.

To this end, suppose that for each product there is a competitive fringe of firms producing the product at marginal cost of $\hat{c} > 0$. Therefore, a retailer can always obtain a product at cost \hat{c} . The fringe is less efficient than either manufacturer, so that in equilibrium manufacturers will supply the retailer(s). However, the fringe constrains the bargaining power of manufacturers. For example, in case of independent retailers, manufacturers cannot extract the full retail profits via the fixed fee. In what follows, we assume that $\hat{c} < w_I^C$,³² which implies that the fringe is not only constraining the fixed fee but also provides a ceiling on the collusive wholesale prices.

³²Since $w_I^C < w_c^C$, this assumption also implies $\hat{c} < w_c^C$.

We demonstrate the robustness of our results by following the analysis with close substitutes provided in Section 6.1. Manufacturers cannot set collusive wholesale prices above \hat{c} if retailers can buy the products from the fringe firms. With independent retailers, colluding manufacturers will therefore optimally set a wholesale price of \hat{c} and a fixed fee of zero. A deviating manufacturer then sets $w_I^D = 0$, thereby inducing his retailer to set a downstream price equal to $\hat{c} - \epsilon$, with $\epsilon \to 0$, and extracting the resulting profit from the retailer.³³ It is easy to see that, as above, Π_I^D is twice as large as Π_I^C , leading to critical discount factor of 1/2.

With a common retailer, manufacturers also optimally collude by setting a wholesale price equal to \hat{c} . The common retailer's outside option of buying from the fringe is then equivalent to the option of rejecting one contract, and manufacturers cannot demand a positive fixed fee. The collusion profit of each manufacturer is then $D^i(p_i(\hat{c},\hat{c}), p_j(\hat{c},\hat{c}))\hat{c}$. Proceeding in the same way as in Section 6.1 to determine whether $\Pi_c^C/\Pi_c^D < 1/2$, we obtain that the equivalent to (26) is

$$D^{i}(p_{i}(0),\infty)) p_{i}(0) > 2D^{i}(p_{i}(\hat{c},\hat{c}),p_{j}(\hat{c},\hat{c})) p_{i}(\hat{c},\hat{c}).$$
(27)

This implies that the analysis is unchanged except for w_c^C replaced by \hat{c} . Since the righthand side of (27) is equal to $D^i(p_i(\hat{c}), \infty)) p_i(\hat{c})$, which is lower than $D^i(p_i(0), \infty)) p_i(0)$, the inequality is fulfilled. As a consequence, the critical discount factor with a common retailer is above the one with independent retailers. Therefore, the intuition driving our result carries over to the case in which retailers have an outside option.

We finally note that the effect becomes weaker the lower \hat{c} . In particular, $D^i(p_i(w), \infty)) p_i(w)$ is falling in w for all w > 0. Since the collusive wholesale price under both channel structures equals \hat{c} , a lower \hat{c} implies an increase in the right-hand side of (27). Therefore, as fringe firms become more efficient, the critical discount factors are getting closer to each other. But, as long as $\hat{c} > 0$, independent retailing facilitates collusion. This result also holds with differentiated products and linear demand.

7 Linear Wholesale Price Contracts

In this section, we study the case in which manufacturers' contract offers consist of a single per-unit wholesale price. That is, franchise fees are not available and the contract proposed by M_i reduces to $C_i(w_i)$. Linear wholesale price contracts are appealing due to their simplic-

³³In fact, any wholesale price between 0 and $\hat{c} - \epsilon$ will lead to the same profit because the deviant can adjust the fixed fee accordingly.

ity and are commonly used as a mechanism governing transactions within the distribution channel in several industries.

7.1 Common Retailer

Downstream Stage

We proceed as in the previous analysis and start with a single downstream firm. Without franchise fees, the common retailer's objective function, when accepting both contracts, is $\pi_c(p_i, p_j) = D^i(p_i, p_j)(p_i - w_i) + D^j(p_j, p_i)(p_j - w_j)$. The first-order conditions for p_i and p_j are identical to those with two-part tariffs specified by (2).³⁴

Upstream Stage and Collusion

In contrast to the equilibrium with two-part tariffs, manufacturers now propose only a wholesale price to the common retailer. Hence, M_i 's maximization program is $\max_{w_i} \Pi_i = D^i(p_i(\mathbf{w}), p_j(\mathbf{w}))w_i$, yielding the system of first-order conditions

$$w_i \left(\frac{\partial D^i(\cdot)}{\partial p_i} \frac{\partial p_i(\mathbf{w})}{\partial w_i} + \frac{\partial D^i(\cdot)}{\partial p_j} \frac{\partial p_j(\mathbf{w})}{\partial w_i} \right) + D^i(\cdot) = 0, \quad i = 1, 2.$$
(28)

The equilibrium wholesale price is above marginal cost since manufacturers can now obtain only profits per unit sold and therefore set a positive margin. With linear demand, the symmetric wholesale price solving the system above is $w_c^N = (\alpha(\beta - \gamma))/(2\beta - \gamma)$. The resulting profit a manufacturer realizes in the punishment phase is

$$\Pi_c^N = \frac{\alpha^2 \beta (\beta - \gamma)}{2(\beta + \gamma)(2\beta - \gamma)^2}.$$
(29)

Now consider collusion. When maximizing joint profits, the manufacturers' optimization program is $\max_{(w_1,w_2)} D^i(p_i(\mathbf{w}), p_j(\mathbf{w}))w_i + D^j(p_j(\mathbf{w}), p_i(\mathbf{w}))w_j$. Maximizing with respect to the wholesale prices gives the following system of first-order conditions:

$$w_i \left(\frac{\partial D^i(\cdot)}{\partial p_i}\frac{\partial p_i(\mathbf{w})}{\partial w_i} + \frac{\partial D^i(\cdot)}{\partial p_j}\frac{\partial p_j(\mathbf{w})}{\partial w_i}\right) + D^i(\cdot) + w_j \left(\frac{\partial D^j(\cdot)}{\partial p_j}\frac{\partial p_j(\mathbf{w})}{\partial w_i} + \frac{\partial D^j(\cdot)}{\partial p_i}\frac{\partial p_i(\mathbf{w})}{\partial w_i}\right) = 0, \ i = 1, 2.$$

With linear demand, the solution to this system is $w_c^C = \alpha/2$. Each manufacturer obtains

³⁴With two-part tariffs, the franchise fees cancel out in the first-order conditions of the common retailer's profit maximization with respect to the retail prices.

an equilibrium profit from collusion given by

$$\Pi_c^C = \frac{\alpha^2}{8(\beta + \gamma)}.$$
(30)

Linear wholesale price contracts impair the ability to coordinate the distribution channel as compared to two-part tariffs: double marginalization occurs in the Nash and the collusion equilibrium. However, in analogy to two-part tariffs, manufacturers set wholesale prices particularly high under collusion, implying that industry profits are relatively low.

Lemma 2. With a common retailer and linear wholesale price contracts, an increase in competition reduces industry profits under manufacturer collusion to a larger extent compared to the stage game equilibrium. However, as competition intensifies, the share of the profits kept by manufacturers remains constant at 2/3, while it decreases in the Nash equilibrium.

With linear wholesale price contracts, neither the collusive regime nor the stage game implies a maximization of the industry profits. It follows from (29) that, analogously to two-part tariffs, the share of the Nash industry profits obtained by manufacturers decreases with the level of competition and vanishes if products become perfect substitutes. This is not the case under collusion. The collusive wholesale prices, and therefore the prices set by the retailer, are independent of the level of competition (i.e., they do not depend on γ). It follows that industry profits decrease as products become closer substitutes. However, similar to the intuition obtained under two-part tariffs, keeping prices constant allows manufacturers to keep up the share of the profits they appropriate from the common retailer.

The above discussion exemplifies that linear contracts do not only fail on channel coordination but also limit the rents that colluding manufacturers can extract from a common retailer. The manufacturers' portion of the industry profits is considerably larger with a two-part tariff.

The next proposition summarizes the results:

Proposition 8. Using linear wholesale price contracts, each manufacturer realizes a collusion profit that is given by (30) if products are distributed through a common retailer. As with two-part tariffs, industry profits under collusion are lower than those under competition, but manufacturers appropriate a larger share.

Deviation and Sustainability of Collusion

Under linear wholesale price contracts, M_i 's maximization program when deviating is $\max_{w_i} D^i(p_i(w_i, w_c^C), p_j(w_c^C, w_i))w_i$. The first-order condition that solves this program is identical to (28) with arguments of p_i and p_j being (w_i, w_c^C) and (w_c^C, w_i) instead of \mathbf{w} , respectively. With linear demand, the solution is $w_c^D = \alpha(2\beta - \gamma)/(4\beta)$, yielding a deviation profit of

$$\Pi_c^D = \frac{\alpha^2 (2\beta - \gamma)^2}{32\beta(\beta + \gamma)(\beta - \gamma)}.$$
(31)

It can be easily verified that the common retailer is better off selling both products if they are sufficiently differentiated. Under wholesale price contracts, the common retailer does not pay a franchise fee to M_j , which renders distribution of product j profitable. However, if products are close substitutes, the demand for M_j 's product becomes zero after a deviation of M_i . Setting $D^j \left(p_j(w_c^C, w_c^D), p_i(w_c^D, w_c^C) \right)$ equal to zero, we obtain that this occurs whenever γ exceeds $(\sqrt{3} - 1)\beta \approx (0.732)\beta$, which is identical to the threshold value $\hat{\gamma}$ derived in Subsection 5.1. If $\gamma > \hat{\gamma}$, the deviant optimally sets a wholesale price of $\hat{w}_c^D = \alpha(2\gamma - \beta)/(2\gamma)$, which drives the rival out of the market.³⁵ The corresponding deviation profit is

$$\widehat{\Pi}_{c}^{D} = \frac{\alpha^{2}(2\gamma - \beta)}{8\gamma^{2}}.$$
(32)

Using (29), (30), (31) and (32), we can determine the critical discount factor δ_c above which manufacturers can sustain collusion under wholesale price contracts if products are sold through a common retailer. We obtain

$$\delta_c = \frac{(2\beta - \gamma)^2}{8\beta (\beta - \gamma) + \gamma^2} \text{ for } \gamma \in [0, \widehat{\gamma}],$$

and

$$\delta_c = \frac{(2\beta - \gamma)^2 \left(\beta^2 - \beta\gamma - \gamma^2\right)}{\beta(2\beta - 3\gamma)(\beta - \gamma)(2\beta + \gamma) - 2\gamma^4} \quad \text{for } \gamma \in (\widehat{\gamma}, \beta).$$
(33)

7.2 Independent Retailers

Consider now that manufacturers distribute through independent retailers and that contracts cannot be observed by the rival retailer. Because the franchise fee does not affect a retailer's maximization program, the retail prices are still determined by (14).

Upstream Stage and Collusion

³⁵As above, \widehat{w}_c^D equals w_c^D at $\gamma = (0.732)\beta$.

In the wholesale stage, the optimization problem of manufacturer M_i is $\max_{w_i} \Pi_i = D^i \left(p_i(w_i, w_j^e), p_j^e \right) w_i$. Differentiating gives

$$w_i \frac{\partial D^i(\cdot)}{\partial p_i} \frac{\partial p_i(w_i, w_j^e)}{\partial w_i} + D^i(\cdot) = 0, \quad i = 1, 2.$$
(34)

With linear demand, the wholesale price that solves the system of first-order conditions, such that retailers' expectations are fulfilled, is $w_I^N = 2\alpha(\beta - \gamma)/(4\beta - 3\gamma)$, which results in a punishment profit of

$$\Pi_I^N = \frac{2\alpha^2\beta(\beta - \gamma)}{(\beta + \gamma)\left(4\beta - 3\gamma\right)^2}.$$
(35)

In case manufacturers collude, their maximization problem is $\max_{w_1,w_2} \sum_{i=1}^2 D^i (p_i(w_i, w_j^e), p_j(w_j, w_i^e)) w_i$, whose solution is determined by the following system of first-order conditions:

$$\left(\frac{\partial D^{i}(\cdot)}{\partial p_{i}}w_{i}+\frac{\partial D^{j}(\cdot)}{\partial p_{i}}w_{j}\right)\frac{\partial p_{i}(w_{i},w_{j}^{e})}{\partial w_{i}}+D^{i}(\cdot)=0, \quad i=1,2.$$

With linear demand, the equilibrium collusive wholesale price is $w_I^C = 2\alpha\beta/(4\beta - \gamma)$, leading to a collusion profit of

$$\Pi_I^C = \frac{2\alpha^2 \beta^2}{(\beta + \gamma)(4\beta - \gamma)^2}.$$
(36)

In contrast to the case with two-part tariffs, manufacturers do no longer maximize industry profits. This is because wholesale price contracts prevent them from channel coordination. Under collusion, manufacturers optimally increase the wholesale price, which softens competition at the retail level. Because their only instrument to extract rents is the per unit margin, they set the collusive wholesale price so that retailers optimally raise the downstream price above the monopoly level. The next proposition summarizes the analysis:

Proposition 9. With independent (exclusive) retailers, each manufacturer realizes a profit from collusion that is given by (36). In contrast to two-part tariffs, collusion profits are below the maximized industry profits.

Deviation and Sustainability of Collusion

If M_i deviates from the collusive agreement, his maximization program is $\max_{w_i} \Pi_i = D^i \left(p_i(w_i, w_I^C), p_j(w_I^C, w_I^C) \right) w_i$. The corresponding first-order condition is identical to (34) except for the arguments of p_i and p_j , which are (w_i, w_I^C) and (w_I^C, w_I^C) , respectively. Solving

for the deviation wholesale price yields $w_I^D = \alpha (2\beta - \gamma)/(4\beta - \gamma)$. The deviant's profit is

$$\Pi_I^D = \frac{\alpha^2 \beta \left(2\beta - \gamma\right)^2}{2(4\beta - \gamma)^2 \left(\beta^2 - \gamma^2\right)}.$$

As above, the demand for the non-deviant's product becomes zero for γ large enough. Inserting w_I^D and w_I^C into M_j 's demand shows that this occurs again if γ is larger than $\hat{\gamma}$. If $\gamma > \hat{\gamma}$, M_i sets a wholesale price equal to $\hat{w}_I^D = 2\alpha\beta(2\gamma - \beta)/(\gamma(4\beta - \gamma))$ to drive M_j 's demand to zero. The resulting deviation profit is

$$\widehat{\Pi}_{I}^{D} = \frac{2\alpha^{2}\beta^{2}(2\gamma - \beta)}{\gamma^{2}(4\beta - \gamma)^{2}}.$$

Determining the critical discount factor δ_I yields

$$\delta_{I} = \frac{(4\beta - 3\gamma)^{2}}{4\beta(4\beta - 5\gamma) + 5\gamma^{2}} \text{ for } \gamma \in [0, \widehat{\gamma}]$$

and

$$\delta_I = \frac{\beta (4\beta - 3\gamma)^2 \left(\beta^2 - \beta\gamma - \gamma^2\right)}{\beta \left(4\beta - 3\gamma\right)^2 \left(\beta^2 - \beta\gamma - \gamma^2\right) - \gamma^5} \quad \text{for} \quad \gamma \in (\widehat{\gamma}, \beta).$$
(37)

7.3 Public Contracts, Stability of Tacit Collusion, and Welfare

Public Contracts

This subsection studies how tacit collusion between manufacturers works if they propose wholesale price contracts to their independent retailers that can be observed by the rival retailer. We delegate all calculations to Appendix B and confine ourselves to a verbal explanation since the formal presentation follows the same lines as the one under private contracts. From subsection 5.2, we know that contract observability involves a strategic effect. In the stage game, manufacturers choose wholesale prices above marginal cost to relax downstream competition. This result extends to the scenario under wholesale price contracts: in the Nash equilibrium, the wholesale price set by manufacturers and their respective stage game profits are higher under public than under private contracts. Turning to collusion, with two-part tariffs the profit is the same under contract observability and secrecy. This is no longer true under wholesale price contracts. The following proposition states how the manufacturers' collusive strategy changes when contracts become observable.

Proposition 10. With independent (exclusive) retailers, each manufacturer sets a lower collusive wholesale price and earns a higher collusion profit under contract observabil-

ity as compared to the scenario under contract secrecy.

Proposition 10 demonstrates that observable contracts lead to higher collusion profits than secret contracts. Interestingly, manufacturers set a collusive wholesale price that is lower under public contracts than under private contracts. The reason for this is rooted in the strategic effect. However, in contrast to the well-known consequence that an increase in wholesale prices dampens downstream competition, the strategic effect induces manufacturers to *lower* their wholesale prices if they collude under wholesale price contracts. The intuition is as follows: colluding manufacturers maximize joint profits at the upstream level. Due to the strategic effect, a decrease in the wholesale price for product i induces retailer jto decrease her retail price because prices are strategic complements. Under wholesale price contracts, a decrease in downstream prices has a beneficial effect on the upstream collusion profit: it reduces double marginalization and leads to an increase in the quantities sold. This is in contrast to the competitive scenario where manufacturers strategically increase wholesale prices to induce the rival retailer to price less aggressively. As a consequence, collusive wholesale prices are higher under secret contracts, which leads to lower quantities, and thus to lower collusion profits.

Stability of Collusion

Following the analysis of Section 6, the natural question is which channel structure is most suitable for manufacturer collusion under wholesale price contracts. Comparing the collusion profit and the critical discount factor under common retailing with those under independent retailing and secret contracts, we obtain that not only the collusion profit but also the critical discount factor is higher under the latter channel structure. However, this does not imply that collusion is more difficult to sustain with independent retailers because manufacturers can also collude only partially in this regime resulting in a collusion profit below the maximum, but making collusion sustainable for a larger range of discount factors. A similar problem arises in the comparison between public and private contracts with independent retailers where, in contrast to two-part tariffs, the collusion profit under observable contracts exceeds the one under private contracts but the critical discount factor is also higher.

In order to provide a meaningful comparison, we need to proceed in a different way than above by focusing on partial collusion. In particular, we equate the collusion profits of the two channel structures that we aim to compare. If we find that one structure yields a strictly lower critical discount factor, this implies, by continuity of the model, that it is possible to find a collusion profit which is slightly higher than under the other structure, and this collusion profit can be sustained for a strictly lower discount factor. The results of this analysis are summarized in the next proposition and illustrated in Figure 3.

Proposition 11. With linear demand, under wholesale price contracts, distributing through a common retailer makes collusion most difficult to sustain. In addition, with independent (exclusive) retailers, public contracts facilitate collusion compared to private contracts. Formally, $\delta_c \geq \delta_I \geq \delta_{IO}$.

Comparing common with independent retailing yields the same qualitative result as with two-part tariffs. In particular, the collusion profit realized under common retailing can be sustained under independent retailing for a strictly larger range of discount factors, no matter if contracts are observable or not. The intuition behind the result slightly differs from to the one with two-part tariffs but follows similar lines. With two-part tariffs, the deviant always induces the common retailer to reject the rival's contract. Under wholesale price contracts, this monopolization is not possible (as long as competition is not too fierce). However, the deviant affects the quantity that the common retailer procures from the rival. When deviating and setting the wholesale price below the collusive level, the deviant induces the common retailer to order less from the rival, which is not possible with independent retailers. As a consequence, the ratio between the deviation and the collusion profit is higher with common retailing compared to independent retailing.

Turning to the comparison between contract observability and secret contracts with independent retailers, we find that the former facilitates collusion for all degrees of product substitutability. This result differs from the one obtained with two-part tariffs where private contracts facilitate collusion if products are close substitutes. The intuition is as follows: the strategic effect allows colluding manufacturers to influence the downstream price of the rival retailer under observable contracts. As explained above, this allows manufacturers to obtain the same collusion profit under public contracts with a lower wholesale price than under private contracts. The reason is that the quantity increase following the reduction in the wholesale prices is larger under public than under private contracts, leading to higher deviation incentives under the latter. As a consequence, observable contracts make tacit collusion more stable than secret contracts if manufacturers distribute through independent retailers.

Welfare

As with two-part tariffs, social welfare is always lower in case that the manufacturers distribute through a common retailer, irrespective of whether contracts are observable or not with independent retailers. This holds for all ranges of the critical discount factor.

In what follows, we focus on private contracts. First, consider that $\delta < \delta_I < \delta_c$, i.e.,



Figure 3: Critical Discount Factors under Partial Collusion ($\beta = 1$)

collusion is not sustainable under both channel structures. We have that $p_c^N = \alpha(3\beta - 2\gamma)/(4\beta - 2\gamma)$ and $p_I^N = 3\alpha(\beta - \gamma)/(4\beta - 3\gamma)$. It is easily checked that $p_c^N > p_I^N$. Second, consider $\delta_I \leq \delta < \delta_c$, i.e., collusion is sustainable with independent retailers but not with a common retailer. The respective downstream price under partial collusion is $p_I^C = \alpha \left(6\beta - \sqrt{2}\sqrt{\beta}\sqrt{\gamma} - 4\gamma\right)/4(2\beta - \gamma)$. A comparison yields $p_c^N - p_I^C = \alpha\sqrt{\beta}\sqrt{\gamma}/\sqrt{2}(4\beta - 2\gamma)$, which is strictly positive. Finally, for $\delta_I < \delta_c \leq \delta$, it directly follows that $p_c^C > p_I^C$ because the collusive downstream price exceeds the Nash downstream price under common retailing. Thus, common retailing always leads to higher consumer prices than independent retailing with public contracts leads to the same qualitative results that for every δ , retail prices are higher under common retailing than under independent retailing.

8 Conclusion

We analyzed the effects of different channel structures on the ability of manufacturers to tacitly collude. We demonstrated that tacit collusion between manufacturers under common retailing works in a fundamentally different way than in case they sell directly to final consumers. In the latter case, manufacturers maximize industry profits under collusion. By contrast, with a common retailer, manufacturers willingly accept industry profits below the static ones. They set higher wholesale prices to increase their profit share at the expense of the retailer, thereby obtaining a larger piece of a smaller pie.

When comparing common with independent retailing, our paper demonstrates that the latter genuinely facilitates collusion. The result holds both for public and private contracts and is robust if manufacturers offer wholesale price contracts instead of a two-part tariff. Deviation under common retailing is shown to be particularly attractive relative to sticking to collusion: a deviating manufacturer monopolizes the downstream market in the period of deviation—an effect not present with independent retailers. However, although independent retailing facilitates collusion, it leads to higher welfare because a common retailer always acts as monopolist in the downstream market.

In case manufacturers sell through independent retailers and coordinate the channel with a two-part tariff, our analysis reveals that secret contracts facilitate collusion if products are close substitutes. This result provides a rationale for why manufacturers want to keep contracts secret. Therefore, it contrasts with the findings obtained in static models, which conclude that manufacturers always prefer public contracts due to their competition-dampening effect at the retail level.

In our analysis we restricted attention to punishments involving infinite reversion of the stage game outcome. A natural question is therefore if our main results still hold with optimal punishment. Characterizing optimal penal codes is difficult in models with differentiated products because manufacturers' profits are positive even during the punishment phase (see e.g., Wernerfelt 1989 or Häckner 1996). Determining the punishment profit then involves the calculation of the optimal punishment length, which cannot be done in closed form. However, with homogeneous goods, optimal penal codes are equivalent to infinite reversion of the stage game because the latter already implies that manufacturers obtain zero profits. Therefore, optimal punishment cannot inflict lower profits on the deviant manufacturer (see e.g., Belleflamme and Peitz 2010). It follows that for homogeneous products, our results hold even under consideration of optimal penal codes. In addition, the intuition of our main result rested on the finding that colluding manufacturers do not maximize industry profits when distributing through a common retailer. This effect is independent of the form of the punishment because it does not affect the punishment phase. This hints to the fact that a similar effect as the one identified in our analysis drives the critical discount factor when considering optimal punishments. However, we leave an analysis of this case for future research.

Finally, our analysis focuses on common versus independent (exclusive) retailing. However, other channel structures are also common. For example, in markets exhibiting shopping costs on the consumer side, competing manufacturers often sell their brands through multiple common retailers. This implies that there is not only competition between the manufacturers (inter-brand competition) but also between retailers (intra-brand competition). A natural extension of our paper for future research would be to analyze whether intra-brand competition leads to new effects on the sustainability of cooperative agreements between suppliers.

Appendix

Appendix A

First, we check whether the common retailer prefers to accept both the offer of the deviant M_i and the one of M_j who sticks to collusion. The demand for product j at the retail level given wholesale prices $w_i = w_c^D$ and $w_j = w_c^C$ is

$$D^{j}(p_{j}(w_{c}^{C}, w_{c}^{D}), p_{i}(w_{c}^{D}, w_{c}^{C})) = \frac{\alpha \left(\beta^{2} - \beta\gamma - \gamma^{2}\right)}{2(\beta - \gamma)(\beta + \gamma)^{2}}.$$

Solving $D^{j}(p_{j}(w_{c}^{C}, w_{c}^{D}), p_{i}(w_{c}^{D}, w_{c}^{C})) = 0$ yields that the demand for M_{j} 's product is only positive if $\gamma \leq (\sqrt{5} - 1)\beta/2 \approx (0.618)\beta$.³⁶ It follows that for $\gamma > (0.618)\beta$, the retailer can never optimally accept both offers. Let us now look at $\gamma \leq (0.618)\beta$. Inserting the wholesale prices $w_{i} = w_{c}^{D}$ and $w_{j} = w_{c}^{C}$ and the franchise fees $T_{i} = T_{c}^{D}$ and $T_{j} = T_{c}^{C}$ into (1) gives

$$\pi_{c}^{D}(w_{c}^{D}, w_{c}^{C}, T_{c}^{D}, T_{c}^{C}) = \frac{\alpha^{2}\gamma \left(2\beta^{2}(\beta - 2\gamma) + \gamma^{2}(2\beta + \gamma)\right)}{4\beta(\beta - \gamma)(\beta + \gamma)^{2}}.$$

The profit of the common retailer when accepting only the deviant's offer is given by 37

$$\pi_c^D(w_c^D, \infty, T_c^D, \infty) = \frac{\alpha^2 \beta \gamma}{2(\beta + \gamma)^3}$$

Subtracting $\pi_c^D(w_c^C, w_c^D, T_c^C, T_c^D)$ from $\pi_c^D(w_c^D, \infty, T_c^D, \infty)$ yields

$$\pi_{c}^{D}(w_{c}^{D},\infty,T_{c}^{D},\infty) - \pi_{c}^{D}(w_{c}^{D},w_{c}^{C},T_{c}^{D},T_{c}^{C}) = \frac{\alpha^{2}\gamma^{2}\left(2\beta^{2}-2\beta\gamma-\gamma^{2}\right)}{4\beta(\beta-\gamma)(\beta+\gamma)^{3}},$$
(38)

which is positive for $\gamma \leq (\sqrt{3} - 1)\beta \approx (0.732)\beta$. Thus, the common retailer prefers to accept only the deviant's offer for all parameter constellations at which the demands for both products can be positive.

Second, let us check whether M_i deviates so that the common retailer only accepts his offer. From the main text, we know that M_i 's profit in this case is given by (12). If M_i deviates so that R_c accepts both offers, his deviation profit maximizes $D^i(w_i, w_c^C)w_i + T_i$ with respect to w_i and T_i . The highest fixed fee that M_i can demand is given by the

³⁶Here and in the following, numbers are rounded to three decimal places.

³⁷As explained in Assumption A.1, the demand function of product *i* if $q_j = 0$ is $D^i(p_i) = (\alpha - p_i)/\beta$. Inserting the optimal retail price for $w_i = w_c^D$ then yields $\pi_c^D(w_c^D, \infty, T_c^D, \infty)$.

expression

$$D^{i}(p_{i}(w_{i}, w_{c}^{C}), p_{j}(w_{c}^{C}, w_{i}))(p_{i}(w_{i}, w_{c}^{C}) - w_{i}) + D^{j}(p_{j}(w_{c}^{C}, w_{i}), p_{i}(w_{i}, w_{c}^{C}))(p_{j}(w_{c}^{C}, w_{i}) - w_{c}^{C}) - T_{i}$$
$$= D^{j}(p_{j}^{M}(w_{c}^{C}), \infty)(p_{j}^{M}(w_{c}^{C}) - w_{c}^{C}).$$

Solving for T_i and inserting into $D^i(p_i(w_i, w_c^C), p_j(w_c^C, w_i))w_i + T_i$, we obtain that M_i deviates by maximizing $D^i(p_i(w_i, w_c^C), p_j(w_c^C, w_i))p_i(w_i, w_c^C) + D^j(p_j(w_c^C, w_i), p_i(w_i, w_c^C))(p_j(w_c^C, w_i) - w_c^C)$ with respect to w_i . With linear demand, the maximization program is

$$\frac{\beta \left(\alpha \beta + w_i(\beta + \gamma)\right) \left(\alpha \beta - w_i(\beta + \gamma)\right)}{4(\beta - \gamma)(\beta + \gamma)^3}$$

Maximizing with respect to w_i yields $w_c^D = 0$, that is, the deviation wholesale price is identical to the one in case R_c accepts only M_i 's offer. Inserting $w_c^D = 0$ yields a deviation profit of

$$\frac{\alpha^2\beta^3}{4(\beta-\gamma)(\beta+\gamma)^3}$$

Subtracting this expression from Π_c^D given by (12) yields

$$\frac{\alpha^2 \gamma^2 \left(2\beta^2 - 2\beta\gamma - \gamma^2\right)}{4\beta (\beta - \gamma)(\beta + \gamma)^3}$$

which is identical to (38). It follows that there does not exist a parameter range where M_i profitably deviates so that both products are distributed. Therefore, the optimal deviation implies that M_i induces R_c to reject M_j 's offer.

Proof of Lemma 1

In the Nash equilibrium, each manufacturer sets $w_c^N = 0$ and R_c sets $p_c^N = \alpha/2$ for either brand. Industry profits are maximized and given by $IP_c^N = \alpha^2/(2(\beta + \gamma))$. In the collusive regime, M_i and M_j set w_c^C above marginal cost (i.e., $w_c^C = \alpha\gamma/(\beta + \gamma)$) and R_c raises the downstream prices to $p_c^C = \alpha(\beta + 2\gamma)/2(\beta + \gamma)$, which is above the monopoly level. Thus, industry profits are below their maximum value and given by $IP_c^C = \alpha^2\beta(\beta + 2\gamma)/2(\beta + \gamma)^3$. The ratio between the industry profits under collusion and those in the Nash equilibrium is $IP_c^C/IP_c^N = \beta(\beta + 2\gamma)/(\beta + \gamma)^2$. Taking the derivative of IP_c^C/IP_c^N with respect to γ yields $\partial (IP_c^C/IP_c^N)/\partial\gamma = -2\beta\gamma/(\beta + \gamma)^3$, which is strictly negative. Therefore, an increase in competition implies a larger decrease in IP_c^C relative to IP_c^N . Using (7) and (9) yields the respective shares of the industry profits obtained by manufacturers, which are $\Pi_c^N/IP_c^N = (\beta - \gamma)/\beta$ in the stage game equilibrium and $\Pi_c^C/IP_c^C = (\beta + \gamma)/(\beta + 2\gamma)$ under collusion. Straightforward calculations show that $\Pi_c^C/IP_c^C > \Pi_c^N/IP_c^N$ and

$$\frac{\partial \frac{\Pi_c^C / IP_c^C}{\Pi_c^N / IP_c^N}}{\partial \gamma} = \frac{2\beta\gamma(2\beta + \gamma)}{(\beta + 2\gamma)^2(\beta - \gamma)^2} > 0.$$

_	

Proof of Proposition 6.

First, using (9) and (17), we can compare the equilibrium collusion profits. This yields $\Pi_c^C - \Pi_I^C = -\gamma \alpha^2 / (4(\beta + \gamma)^2) < 0$. Because $\Pi_{IO}^C = \Pi_I^C$, it follows that $\Pi_{IO}^C > \Pi_c^C$. Thus, collusion profits are strictly larger under independent than under common retailing.

Second, using (13) and (20), we can compare the critical discount factor above which manufacturers can sustain collusion under common retailing with the one under independent retailing and private contracts. Doing so yields

$$\delta_c - \delta_I = \frac{8\beta^2(\beta - \gamma) - \gamma^2(\beta + \gamma)}{2(2\beta + \gamma)\left(8\beta(\beta - \gamma) + \gamma^2\right)} \quad \text{for } \gamma \in [0, \widehat{\gamma}]$$

and

$$\delta_c - \delta_I = \frac{\beta \left(4\beta^3(\beta - \gamma) + \gamma^2 \left(\gamma(4\beta + 5\gamma) - 7\beta^2\right)\right)}{2(2\beta + \gamma) \left(4\beta^3(2\gamma - \beta) - \gamma^2 \left(\beta^2 + 3\gamma\beta - 2\gamma^2\right)\right)} \quad \text{for} \quad \gamma \in (\widehat{\gamma}, \beta).$$

It can be immediately verified that the first expression is positive if and only if $\gamma \leq (0.838)\beta$. Since the expression is valid for $\gamma \leq \hat{\gamma} \equiv (0.732)\beta$, it is positive for $\gamma \in [0, \hat{\gamma}]$. The numerator of the second expression is positive for $\gamma \in [0, \beta)$ and the denominator is positive for $\gamma \in$ $((0.591)\beta, \beta]$. Since the second expression is valid for $\gamma > (0.732)\beta$, it is strictly positive for $\gamma \in (\hat{\gamma}, \beta)$. Finally, comparing the critical discount factor under common retailing with the one under independent retailing and observable contracts yields that $\delta_c - \delta_{IO}$ is given by

$$\frac{32\beta^4(\beta-\gamma)-4\beta^2\gamma^2(5\beta-4\gamma)+\gamma^4(3\beta-\gamma)}{2(2\beta+\gamma)\left(32\beta^3(\beta-\gamma)-12\beta\gamma^2(\beta-\gamma)+\gamma^4\right)} \text{ for } \gamma \in [0,\widetilde{\gamma}],$$

$$\frac{64\beta^7(\beta-\gamma)-16\beta^5\gamma^2(9\beta-8\gamma)+28\beta^3\gamma^4(4\beta-3\gamma)-25\gamma^6\beta^2+3\gamma^7(4\beta+\gamma)}{2(2\beta+\gamma)\left(64\beta^7(2\gamma-\beta)+16\beta^4\gamma^2(\beta-9\gamma)+4\beta^2\gamma^4(12\beta+7\gamma)-7\beta\gamma^6-3\gamma^7\right)} \text{ for } \gamma \in (\widetilde{\gamma},\beta).$$

Both expressions are positive in the relevant range, implying that $\delta_c > \delta_{IO}$. Consequently, collusion is easier to sustain with independent retailers, regardless of whether contracts are observable or not.

Proof of Proposition 7.

We use (20) and (25) to compare the critical discount factor under private contracts with the one under public contracts: the difference $\delta_I - \delta_{IO}$ is given by

$$\frac{4\beta\gamma^{3}(4\beta-3\gamma)(\beta-\gamma)}{[8\beta^{2}-8\beta\gamma+\gamma^{2}]\left[\gamma^{4}+4\beta(\beta-\gamma)(8\beta^{2}-3\gamma^{2})\right]} \text{ for } \gamma \in [0,\widehat{\gamma}],$$

$$\frac{\gamma^{8}+4\beta(\beta-\gamma)(2\beta-\gamma)\left(8\beta^{5}-12\beta^{4}\gamma-8\beta^{3}\gamma^{2}+10\beta^{2}\gamma^{3}+4\beta\gamma^{4}-\gamma^{5}\right)}{\kappa\left(4\beta^{3}(\beta-2\gamma)+\beta\gamma^{2}(\beta+3\gamma)-2\gamma^{4}\right)\left(4\beta(8\beta^{2}-3\gamma^{2})(\beta-\gamma)+\gamma^{4}\right)} \text{ for } \gamma \in (\widehat{\gamma},\widetilde{\gamma}],$$

$$\frac{\gamma^{4}\left(\beta-\gamma\right)\left(4\beta^{2}-4\beta\gamma-\gamma^{2}\right)\left(12\beta^{4}+4\beta^{3}\gamma-15\beta^{2}\gamma^{2}+3\gamma^{4}\right)}{\kappa\left(64\beta^{7}(2\gamma-\beta)+16\beta^{4}\gamma^{2}(\beta-9\gamma)+4\beta^{2}\gamma^{4}(12\beta+7\gamma)-\gamma^{6}(7\beta+3\gamma)\right)} \text{ for } \gamma \in (\widetilde{\gamma},\beta).$$

with $\kappa \equiv (4\beta^3(\beta - 2\gamma) + \beta\gamma^2(\beta + 3\gamma) - 2\gamma^4)$. It is readily checked that the first term is positive for $\gamma \in [0, \beta)$. The second expression is valid between $\hat{\gamma} = (0.732)\beta$ and $\tilde{\gamma} = (0.841)\beta$. Setting it equal to zero and solving for γ , we obtain that it is positive for $\gamma \leq (0.825)\beta$ and negative for $\gamma > (0.825)\beta$. Finally, it is easy to verify that the third expression is strictly negative for all γ between $\tilde{\gamma}$ and β . Therefore, collusion is easier to sustain under public than under private contracts if $\gamma \leq (0.825)\beta$ and vice versa.

Proof of Lemma 2

In the Nash equilibrium, manufacturers set $w_c^N = \alpha(\beta - \gamma)(2\beta - \gamma)$ and R_c sets $p_c^N = \alpha(3\beta - 2\gamma)/2(2\beta - \gamma)$. Industry profits are given by

$$IP_c^N = \frac{\alpha^2 \beta (3\beta - 2\gamma)}{2(2\beta - \gamma)^2 (\beta + \gamma)}.$$

Under collusion, M_i and M_j raise the wholesale price to $w_c^C = \alpha/2$, resulting in a downstream price of $p_c^C = 3\alpha/4$. Industry profits are given by $IP_c^C = 3\alpha^2/8(\beta + \gamma)$.

Following the proof of Lemma 1, the ratio between industry profits under collusion and in the Nash equilibrium is $IP_c^C/IP_c^N = 3(2\beta - \gamma)^2/4\beta(3\beta - 2\gamma)$. Taking the derivative with respect to γ yields $\partial \left(IP_c^C/IP_c^N\right)/\partial\gamma = -3(2\beta^2 - 3\beta\gamma + \gamma^2)/2\beta(3\beta - 2\gamma)^2$, which is strictly negative. The portion of the industry profits obtained by manufacturers are derived from (29) and (30) and given by $\Pi_c^N/IP_c^N = 2(\beta - \gamma)/(3\beta - 2\gamma)$ in the stage game equilibrium and by $\Pi_c^C/IP_c^C = 2/3$ under collusion. As with two-part tariffs, $\Pi_c^C/IP_c^C > \Pi_c^N/IP_c^N$ and

$$\frac{\partial \frac{\Pi_c^C/IP_c^C}{\Pi_c^N/IP_c^N}}{\partial \gamma} = \frac{\beta}{3(\beta - \gamma)^2} > 0.$$

Appendix B

Under wholesale price contracts and contract observability, the stage game wholesale price that manufacturers set with linear demand is $w_{IO}^N = \alpha(\beta - \gamma)(2\beta + \gamma)/(4\beta^2 - \beta\gamma - 2\gamma^2)$. This wholesale price exceeds the one under private contracts, that is,

$$w_{IO}^N - w_I^N = \frac{\alpha \gamma^2 (\beta - \gamma)}{(4\beta - 3\gamma) \left(\beta (4\beta - \gamma) - 2\gamma^2\right)} > 0.$$

The resulting profit in the punishment phase is

$$\Pi_{IO}^{N} = \frac{\alpha^{2}\beta(\beta-\gamma)(2\beta+\gamma)(2\beta^{2}-\gamma^{2})}{(4\beta^{2}-\beta\gamma-2\gamma^{2})^{2}(2\beta^{2}+\beta\gamma-\gamma^{2})}.$$

As with two-part tariffs, the stage game profit under observable contracts is larger than the one under private contracts given by (35). We have that

$$\Pi_{IO}^{N} - \Pi_{I}^{N} = \frac{\alpha^{2}\beta\gamma^{3}(\beta-\gamma)\left(2\beta(2\beta-\gamma)-\gamma^{2}\right)}{(4\beta-3\gamma)^{2}(2\beta-\gamma)(\beta+\gamma)\left(\beta(4\beta-\gamma)-2\gamma^{2}\right)^{2}},$$

which is strictly positive. The equilibrium wholesale price chosen by manufacturers under collusion is $w_{IO}^C = \alpha/2$. This collusive wholesale price is below the one under private contracts, i.e., $w_{IO}^C - w_I^C = -\alpha\gamma/(8\beta - 2\gamma) < 0$. The respective collusion profit is given by

$$\Pi_{IO}^C = \frac{\alpha^2 \beta}{2(\beta + \gamma)(2\beta - \gamma)}.$$

Subtracting the collusion profit under private contracts (36) gives

$$\Pi_{IO}^C - \Pi_I^C = \frac{\alpha^2 \beta \left(4\beta(2\beta - \gamma) + \gamma^2\right)}{2(2\beta - \gamma)(\gamma - 4\beta)^2(\beta + \gamma)},$$

which is positive. Thus, when colluding, manufacturers generate higher profits under observable contracts than under secret contracts. If M_i deviates from the collusive agreement, M_i optimally sets $w_{IO}^D = \alpha \left(4\beta^2 - \beta\gamma - 2\gamma^2\right)/4 \left(2\beta^2 - \gamma^2\right)$. The deviation profit is given by

$$\Pi_{IO}^{D} = \frac{\alpha^{2}\beta \left(-4\beta^{2}+\beta\gamma+2\gamma^{2}\right)^{2}}{16(\beta-\gamma)(2\beta-\gamma)(\beta+\gamma)(2\beta+\gamma)\left(2\beta^{2}-\gamma^{2}\right)}.$$

As above, the demand for the product of the non-deviant becomes zero for γ large enough. Inserting w_{IO}^D and w_{IO}^C into M_j 's demand function shows that this occurs if $\gamma > (0.887)\beta$. If $\gamma > (0.887)\beta$, M_i sets a wholesale price equal to $\widehat{w}_{IO}^D = \alpha \left(\gamma(2\beta + \gamma) - 2\beta^2\right)/2\beta\gamma$, which drives M_j 's demand to zero. The resulting deviation profit is

$$\widehat{\Pi}^{D}_{IO} = \frac{\alpha^2 \left(\gamma (2\beta + \gamma) - 2\beta^2\right)}{4\beta\gamma^2}$$

Determining the critical discount factor δ_{IO} yields

$$\delta_{IO} = \frac{\left(4\beta^2 - \beta\gamma - 2\gamma^2\right)^2}{16\beta^3(2\beta - \gamma) - 31\beta^2\gamma^2 + 8\gamma^3(\beta + \gamma)}$$

for $\gamma \leq (0.887)\beta$ and

$$\delta_{IO} = \frac{(4\beta^2 - \beta\gamma - 2\gamma^2)^2 (2\beta^3 (2\beta - \gamma) - \beta\gamma^2 (5\beta - \gamma) + \gamma^4)}{64\beta^7 (\beta - \gamma) - 31\beta^2 \gamma^6 - 2\beta^4 \gamma^2 (62\beta^2 - 45\gamma^2) + 51\beta^3 \gamma^3 (2\beta^2 - \gamma^2) + 4\gamma^7 (2\beta + \gamma)}$$

for $\gamma \in ((0.887)\beta, \beta)$.

Appendix C

Sustainability of Collusion: Common vs. Independent Retailing under Private Contracts

To derive comparable results we need to consider partial collusion so that the collusion profit with independent retailers and private contracts is the same as the one with a common retailer given by (30). Determining the symmetric wholesale price that gives manufacturers the same collusion profit as under common retailing, we obtain

$$w_I^C = \frac{\alpha \left(2\beta - \sqrt{2}\sqrt{\beta}\sqrt{\gamma}\right)}{4\beta}.$$

If M_i deviates from this partially collusive agreement, his maximization program is $\max_{w_i} \prod_i = D^i \left(p_i(w_i, w_I^C), p_j(w_I^C, w_I^C) \right) w_i$. The corresponding first-order condition is identical to (34) except for the arguments of p_i and p_j , which are (w_i, w_I^C) and (w_I^C, w_I^C) , respectively. With linear demand, M_i sets a deviation wholesale price which is given by

$$w_I^D = \frac{\alpha \left(2\beta (4\beta - 3\gamma) - \gamma \sqrt{2}\sqrt{\beta}\sqrt{\gamma}\right)}{8\beta (2\beta - \gamma)},$$

yielding a profit from deviation equal to

$$\Pi_I^D = \frac{\alpha^2 \left(8\beta\sqrt{\beta} - 6\gamma\sqrt{\beta} - \gamma\sqrt{2}\sqrt{\gamma}\right)^2}{128 \left(\beta^2 - \gamma^2\right) \left(2\beta - \gamma\right)^2}.$$

Proceeding in the familiar way shows that this deviation implies that the demand of M_j is positive if $\gamma \leq (0.970)\beta$. If $\gamma > (0.970)\beta$, M_i chooses a deviation wholesale price so that the demand for M_j 's product is zero. The respective deviation profit is

$$\widehat{\Pi}_{I}^{D} = \frac{\alpha^{2} \left(2\sqrt{\beta} + \sqrt{2}\sqrt{\gamma}\right) \left(4\beta\sqrt{\beta}(2\gamma - \beta) + \sqrt{2}\sqrt{\gamma} \left(\gamma^{2} - 2\beta^{2}\right) - 2\gamma^{2}\sqrt{\beta}\right)}{16\gamma^{2}(2\beta - \gamma)^{2}},$$

Before determining the critical discount factor, we compare the difference between the Nash profit with independent retailers (35) and the collusion profit with a common retailer (30). It can be easily shown that the former exceeds the latter if $\gamma \leq (0.889)\beta$. This implies that the the collusion profit with a common retailer can be sustained for all $\delta \geq 0$ under independent retailing.

Turning to the range $\gamma > (0.889)\beta$, the critical discount factor δ_I above which manufacturers can sustain collusion under independent retailing is

$$\delta_I = \frac{(4\beta - 3\gamma)^2 \left(3\sqrt{2}\sqrt{\gamma} - 4\sqrt{\beta}\right)}{\sqrt{\gamma} \left(32\beta \left(\sqrt{2}(\beta - \gamma) - \sqrt{\beta}\right) + \gamma^2 \left(28\sqrt{\beta}\sqrt{\gamma} + 3\sqrt{2}\right)\right)},$$

for $\gamma \in [(0.889)\beta, (0.970)\beta]$, and

$$\delta_{I} = \frac{\beta(4\beta - 3\gamma)^{2} \left(2\sqrt{\beta} \left(\beta\sqrt{\beta} \left(2 \left(\beta + \sqrt{2}\sqrt{\beta}\sqrt{\gamma}\right) - \gamma\right) - 2\gamma^{2}\sqrt{2}\sqrt{\gamma}\right)\right)}{\beta \left(2\sqrt{\beta} \left(2\sqrt{2}\sqrt{\gamma}(\beta - \gamma)(\beta + \gamma)(4\beta - 3\gamma)^{2} + \sqrt{\beta} \Phi\right) + 17\gamma^{5}\right) - 9\gamma^{6}},$$

for $\gamma \in ((0.970)\beta, \beta)$, with $\Phi = 32\beta^3(\beta - 2\gamma) + 2\gamma^2(21\beta^2 + \gamma^2) - 17\beta\gamma^3$.

We know that for $\gamma \leq (0.889)\beta$, Π_I^N is higher than Π_c^C . Thus, to provide a meaningful comparison between δ_I and δ_c we focus the range $\gamma > (0.889)\beta$. We must distinguish between two cases. We start with the case $(0.889)\beta < \gamma \leq (0.970)\beta$. Subtracting the critical discount factor with independent retailers and private contracts from the one with a common retailer given by (33), we obtain that $\delta_c - \delta_I$ is given by

$$\frac{1}{2} + \frac{\sqrt{2}\sqrt{\beta}}{\sqrt{\gamma}} + \frac{\gamma^4}{\psi} + \frac{8\sqrt{\beta}\sqrt{\gamma}\left(7\sqrt{2}\sqrt{\beta}\sqrt{\gamma} - 12\beta + 11\gamma\right) - 51\sqrt{2}\gamma^2}{8\sqrt{\beta}\left(8\sqrt{\beta}\left(\sqrt{2}\beta - \sqrt{\beta}\sqrt{\gamma} - \sqrt{2}\gamma\right) + 7\gamma\sqrt{\gamma}\right) + 6\sqrt{2}\gamma^2}$$

with $\psi = \beta(2\beta - 3\gamma)(\beta - \gamma)(2\beta + \gamma) - 2\gamma^4$. Setting this expression equal to zero and solving for γ reveals that there is no solution for $\gamma \in [(0.889)\beta, (0.970)\beta]$. The two solutions closest to this range are $\gamma = (0.607)\beta$ and $\gamma = (0.982)\beta$, and for all γ in between, the expression is positive.

In the second case, i.e., for $(0.970)\beta < \gamma < \beta$, $\delta_c - \delta_I$ yields

$$\frac{\beta(4\beta - 3\gamma)^2 \left(3\gamma^3 - 2\sqrt{\beta} \left(\beta\sqrt{\beta} \left(2\sqrt{\beta} \left(\sqrt{\beta} + \sqrt{2}\sqrt{\gamma}\right) - \gamma\right) + 2\gamma^2\sqrt{2}\sqrt{\gamma}\right)\right)}{\beta \left(2\sqrt{\beta} \left(2\sqrt{2}\sqrt{\gamma}(\beta - \gamma)(\beta + \gamma)(4\beta - 3\gamma)^2 + \sqrt{\beta}\Phi\right) + 17\gamma^5\right) - 9\gamma^6} + \frac{(\beta^2 - \beta\gamma - \gamma^2)(2\beta - \gamma)^2}{\psi}.$$

Proceeding as before shows that this expression is positive for $\gamma \in [(0.970)\beta, \beta)$.

Sustainability of Collusion under Independent Retailing: Private vs. Public Contracts

To compare sustainability of collusion between private and public contracts, we proceed as above and consider partial collusion. When manufacturers collude so that the collusion profit with public contracts is identical to Π_I^C , given by (36), each of them sets a wholesale price of $w_{IO}^C = \alpha (2\beta - \gamma)/(4\beta - \gamma)$.

When deviating from the partially collusive agreement, the deviant M_i optimally sets $w_{IO}^D = \alpha \left(4\beta \left(2\beta^2 - \beta\gamma - \gamma^2\right) + \gamma^3\right) / \left(2(4\beta - \gamma)(2\beta^2 - \gamma^2)\right)$, realizing a deviation profit given by

$$\Pi_{IO}^{D} = \frac{\alpha^{2}\beta \left(4\beta\gamma^{3}(\beta-\gamma)(2\beta+\gamma)\right)^{2}}{4(4\beta-\gamma)^{2} \left(8\beta^{6}-7\beta^{2}\gamma^{2}\left(2\beta^{2}-\gamma^{2}\right)-\gamma^{6}\right)}.$$

 M_i 's deviation implies that M_j 's demand becomes zero if $\gamma > (0.940)\beta$. It is then optimal for M_i to deviate by setting $\widehat{w}_{IO}^D = \alpha \left(4\beta(\gamma - \beta) + \gamma^2\right) / (\gamma(4\beta - \gamma))$, which keeps M_j out of the market. M_i 's resulting profit is given by

$$\widehat{\Pi}^{D}_{IO} = \frac{2\alpha^{2}\beta\left(4\beta(\gamma-\beta)+\gamma^{2}\right)}{\gamma^{2}(4\beta-\gamma)^{2}}$$

Determining the critical discount factor δ_{IO} in the standard way gives

$$\frac{\gamma^2 \left(4\beta^2 - \beta\gamma - 2\gamma^2\right)^2}{\beta \left(8\beta^2 (2\beta - \gamma) \left(4\beta^2 - \beta\gamma - 4\gamma^2\right) + \gamma^4 (13\beta - 4\gamma)\right)}$$

for $0 < \gamma \leq (0.940)\beta$ and

$$\frac{2\left(\beta(4\beta-\gamma)-2\gamma^2\right)^2\left(2\beta\left(2\beta\left(\beta(2\beta-\gamma)-2\gamma^2\right)+\gamma^3\right)+\gamma^4\right)}{2\beta^2\gamma^3\left(2\beta\left(94\beta^2-43\gamma^2\right)-41\gamma^3\right)+16\beta^4\left(16\beta^3(\beta-\gamma)-\gamma^2\left(27\beta^2-17\gamma^2\right)\right)+\gamma^7(25\beta+9\gamma)}$$

for $\gamma \in ((0.940)\beta, \beta).^{38}$

Subtracting this critical discount factor from the one under private contracts given by (37) yields that $\delta_I - \delta_{IO}$ equals

$$\frac{4(\beta-\gamma)\left(2\beta(2\beta-\gamma)-\gamma^2\right)\left(4\beta^3\left(4\beta(4\beta-3\gamma)-13\gamma^2\right)+5\gamma^3\left(7\beta^2-\gamma^2\right)+11\beta\gamma^4\right)}{\beta\left(4\beta(4\beta-5\gamma)+5\gamma^2\right)\left(8\beta^2(2\beta-\gamma)\left(4\beta^2-\beta\gamma-4\gamma^2\right)+\gamma^4(13\beta-4\gamma)\right)}$$

for $\gamma \in [0, \widehat{\gamma}]$,

$$\frac{17}{4} + \frac{\gamma}{\beta} + \frac{\Xi}{\Xi - \gamma^5} + \frac{\beta \left(8\beta(4\beta - \gamma)\left(9\gamma(\beta + 4\gamma) - 34\beta^2\right) - 289\gamma^4\right)}{4 \left(\gamma^4(13\beta - 4\gamma) + 8\beta^2 \left(4\beta^2 - \beta\gamma - 4\gamma^2\right)(2\beta - \gamma)\right)}$$

for $\gamma \in (\widehat{\gamma}, (0.940)\beta]$, and

$$\frac{\gamma^{5}(\beta-\gamma)\left(64\beta^{5}(\beta+\gamma)(4\beta-5\gamma)-4\gamma^{3}\left(9\beta^{4}+2\gamma^{4}\right)+\beta\gamma^{4}\left(153\beta^{2}+49\beta\gamma-23\gamma^{2}\right)\right)}{\left(256\beta^{7}(\beta-\gamma)-2\beta^{2}\gamma^{5}(86\beta+41\gamma)-8\beta^{4}\gamma^{2}\left(54\beta^{2}-47\beta\gamma-34\gamma^{2}\right)+25\beta\gamma^{7}+9\gamma^{8}\right)\left(\Xi-\gamma^{5}\right)}$$

for $\gamma \in ((0.940)\beta, \beta]$, with $\Xi = \beta (\beta^2 - \beta\gamma - \gamma^2) (4\beta - 3\gamma)^2$. It can be shown that all three expressions are positive for the relevant parameter ranges. This implies that $\delta_I > \delta_{IO}$.

Sustainability of Collusion: Common vs. Independent Retailing under Public Contracts

The proof proceeds along the same lines as those above. First, it is easily shown that $\Pi_{IO}^{N} > \Pi_{c}^{C}$ if $\gamma \leq (0.926)\beta$. We then determine the partially collusive wholesale price (so that $\Pi_{IO}^{C} = \Pi_{c}^{C}$) and the corresponding optimal deviation profit for $\gamma > (0.926)\beta$. We can then determine the critical discount factor under independent retailing and observable contracts. We have that δ_{IO} is given by

$$\frac{\left(4\beta-\beta\gamma-2\gamma^2\right)^2\left(2\beta+\gamma\right)\left(8\beta^3-7\beta^2\gamma-4\beta\gamma^2+4\gamma^3\right)-\Upsilon}{\beta\left(\beta\gamma(2\beta+\gamma)\left(8\beta^3(4\beta-3\gamma)-23\beta^2\gamma^2+4\gamma^3(3\beta+\gamma)-\Upsilon\right)\right)},$$

with $\Upsilon = 2\sqrt{2}\sqrt{\beta\gamma} (4\beta - \beta\gamma - 2\gamma^2)^3$, if $(0.926)\beta < \gamma \le (0.980)\beta$, and

$$\frac{(4\beta - \beta\gamma - 2\gamma^2)^2 (2\beta + \gamma) (4\beta^4 + 2\beta^3\gamma - 3\beta^2\gamma^2 - 3\beta\gamma^3 - \gamma^4) + \upsilon}{64\beta^6 (2\beta^3 - \beta^2\gamma + \gamma^3) - 2\beta^4\gamma^2 (124\beta^3 - 3\gamma^3 - 91\beta\gamma^2) - 63\beta^3\gamma^6 - 19\beta^2\gamma^7 + 4\gamma^8 (2\beta + \gamma) + \upsilon}$$

with $v = \Upsilon(\beta + \gamma)(\beta - \gamma)(2\beta + \gamma)(2\beta - \gamma)$, for $\gamma \in ((0.980)\beta, \beta)$. We obtain that $\delta_c > \delta_{IO}$ for all $\gamma \in ((0.926)\beta, \beta)$.

 $^{^{38}\}mathrm{Note}$ that the Nash profit under public contracts is smaller than the collusion profit under private contracts.

References

- Athey, S., K. Bagwell. 2001. Optimal collusion with private information. *RAND Journal* of *Economics* 32(3) 428-465.
- Belleflamme, P., M. Peitz. 2010. Industrial organization: Markets and strategies. Cambridge University Press, Cambridge UK.
- Bernheim, B.D., M.D. Whinston. 1985. Common marketing agency as a device for facilitating collusion. RAND Journal of Economics 16(2) 269-281.
- Bernheim, B.D., M.D. Whinston. 1998. Exclusive dealing. *Journal of Political Economy* 106(1) 1041-1080.
- Besanko, D, M.K. Perry. 1993. Equilibrium incentives for exclusive dealing in a differentiated products oligopoly. *RAND Journal of Economics* 24(4) 646-667.
- Bonanno, G., J. Vickers. 1988. Vertical separation. *Journal of Industrial Economics* 36(3) 257-265.
- Bresnahan, T.F. 1987. Competition and collusion in the American automobile industry: The 1955 price war. *Journal of Industrial Economics* 35(4) 457-482.
- Cachon, G.P., A.G. Kök. 2010. Competing manufacturers in a retail supply chain: On contractual form and coordination. *Management Science* 56(3) 571-589.
- Caillaud, B., P. Rey. 1995. Strategic aspects of vertical delegation. European Economic Review 39(3-4) 421-431.
- Choi, S.C. 1991. Price competition in a channel structure with a common retailer. *Marketing Science* 10(4) 271-296.
- Coughlan, A. 1985. Competition and cooperation in marketing channel choice: Theory and application. *Marketing Science* 4(2) 110-129.
- Coughlan, A., B. Wernerfelt. 1989. On credible delegation by oligopolists: A discussion of distribution channel management. *Management Science* 35(2) 226-239.
- European Commission. 2010. Guidelines on vertical restraints. Commission notice, Brussels, http://ec.europa.eu/competition/antitrust/legislation/guidelines_vertical_en.pdf.
- Gal-Or, E. 1991. A common agency with incomplete information. *RAND Journal of Economics* 22(2) 274-286.
- Gabrielsen, T.S. 1997. Equilibrium retail distribution systems. International Journal of Industrial Organization 16(1) 105-120.

- Gilo, D., Y. Yehezkel. 2016. Dynamic vertical collusion with secret contracts. Working paper, Tel-Aviv University Recanati School of Business, Tel-Aviv.
- Häckner, J. 1996. Optimal symmetric punishments in a bertrand differentiated products oligopoly. *International Journal of Industrial Organization* 14(5) 611-630.
- Jullien, B., P. Rey. 2007. Resale price maintenance and collusion. RAND Journal of Economics 38(4) 983-1001.
- Katz, M. 1991. Game-playing agents: Unobservable contracts as precommitment. RAND Journal of Economics 22(3) 307-328.
- Lin, Y.J. 1990. The dampening-of-competition effect of exclusive dealing. *Journal of Industrial Economics* 39(2) 209-223.
- Martimort, D. 1996. Exclusive dealing, common agency, and multiprincipals incentive theory. *RAND Journal of Economics* 27(1) 1-31.
- Maxton, G.P., J. Wormald. 2004. *Time for a model change. Re-engineering the global automotive industry.* Cambridge University Press, Cambridge UK.
- McGuire, T.W., R. Staelin. 1983. An industry equilibrium analysis of downstream vertical integration. *Marketing Science* 2(2) 161-191.
- Moorthy, S. 1988. Strategic decentralization in channels. *Marketing Science* 7(4) 335-355.
- Nocke, V., L. White. 2007. Do vertical mergers facilitate upstream collusion? *American Economic Review* 97(4) 1321-1339.
- Nocke, V., L. White. 2010. Vertical mergers, collusion, and disruptive buyers. *International Journal of Industrial Organization* 28(4) 350-354.
- Normann, H.T. 2009. Vertical integration, raising rivals' costs and upstream collusion. European Economic Review 53(4) 461-480.
- O'Brien, D.P., G. Shaffer. 1993. On the dampening-of-competition effect of exclusive dealing. *Journal of Industrial Economics* 41(2) 215-221.
- O'Brien, D.P., G. Shaffer. 1997. Nonlinear supply contracts, exclusive dealing, and equilibrium market foreclosure. *Journal of Economics and Management Strategy* 6(2) 755-785.
- Pagnozzi, M., S. Piccolo. 2011. Vertical separation with private contracts. The Economic Journal 122(March) 173-207.
- Palmatier, R.W., L.W. Stern, A.I. El-Ansary. 2016. Marketing Channel Strategy Routledge, New York, USA.

- Panteva, N. 2011. Cosmetics & beauty products manufacturing in the US. IBISWorld Industry Report 32562.
- Piccolo, S., M. Reisinger. 2011. Exclusive territories and manufacturers' collusion. Management Science 57(7) 1250-1266.
- Piccolo, S., J. Miklòs-Thal. 2012. Colluding through suppliers. RAND Journal of Economics 43(3) 492-513.
- Plunkett, J.W. 2012. Plunkett's Automobile Industry Almanac. Plunkett Research, Houston, TX.
- Rey, P., J. Stiglitz. 1995. The role of exclusive territories in producers' competition. RAND Journal of Economics 26(3) 432-451.
- Saltzman, H., R. Levy, J.C. Hilke. 1999. Transformation and continuity: The U.S. carbonated soft drink bottling industry and antitrust policy since 1980. Bureau of Economics Staff Report. Federal Trade Commission.
- Schinkel, M.P., J. Tuinstra, J. Rüggeberg. 2008. Illinois walls: How barring indirect purchaser suits facilitates collusion. *RAND Journal of Economics* 39(3) 683-698.
- Sudhir, K. 2001a. Competitive pricing behavior in the auto market: A structural analysis. Marketing Science 20(1) 42-60.
- Sudhir, K. 2001b. Structural analysis of manufacturer pricing in the presence of a strategic retailer. *Marketing Science* 20(3) 244-264.
- Tirole, J. 1988. The Theory of Industrial Organization. MIT Press, Cambridge, MA.
- Trefis Analysis. 2011. L'Oreal: Acquire to survive? *Forbes* (January 2011), http://www.forbes.com/sites/greatspeculations/2011/01/18/loreal-acquire-to-survive/.
- Wernerfelt, B. 1989. Tacit collusion in differentiated cournot games. *Economics Letters* 29(4) 303-306.

PREVIOUS DISCUSSION PAPERS

- 261 Reisinger, Markus and Thomes, Tim Paul, Manufacturer Collusion: Strategic Implications of the Channel Structure, July 2017.
- 260 Shekhar, Shiva and Wey, Christian, Uncertain Merger Synergies, Passive Partial Ownership, and Merger Control, July 2017.
- 259 Link, Thomas and Neyer, Ulrike, Friction-Induced Interbank Rate Volatility under Alternative Interest Corridor Systems, July 2017.
- 258 Diermeier, Matthias, Goecke, Henry, Niehues, Judith and Thomas, Tobias, Impact of Inequality-Related Media Coverage on the Concerns of the Citizens, July 2017.
- 257 Stiebale, Joel and Wößner, Nicole, M&As, Investment and Financing Constraints, July 2017.
- 256 Wellmann, Nicolas, OTT-Messaging and Mobile Telecommunication: A Joint Market? An Empirical Approach, July 2017.
- 255 Ciani, Andrea and Imbruno, Michele, Microeconomic Mechanisms Behind Export Spillovers from FDI: Evidence from Bulgaria, June 2017.
- 254 Hunold, Matthias and Muthers, Johannes, Capacity Constraints, Price Discrimination, Inefficient Competition and Subcontracting, June 2017.
- 253 Dertwinkel-Kalt, Markus and Köster, Mats, Salient Compromises in the Newsvendor Game, June 2017. Forthcoming in: Journal of Economic Behavior and Organization.
- 252 Siekmann, Manuel, Characteristics, Causes, and Price Effects: Empirical Evidence of Intraday Edgeworth Cycles, May, 2017.
- 251 Benndorf, Volker, Moellers, Claudia and Normann, Hans-Theo, Experienced vs. Inexperienced Participants in the Lab: Do they Behave Differently?, May 2017. Forthcoming in: Journal of the Economic Science Association.
- 250 Hunold, Matthias, Backward Ownership, Uniform Pricing and Entry Deterrence, May 2017.
- 249 Kreickemeier, Udo and Wrona, Jens, Industrialisation and the Big Push in a Global Economy, May 2017.
- 248 Dertwinkel-Kalt, Markus and Köster, Mats, Local Thinking and Skewness Preferences, April 2017.
- 247 Shekhar, Shiva, Homing Choice and Platform Pricing Strategy, March 2017.
- 246 Manasakis, Constantine, Mitrokostas, Evangelos and Petrakis, Emmanuel, Strategic Corporate Social Responsibility by a Multinational Firm, March 2017.
- 245 Ciani, Andrea, Income Inequality and the Quality of Imports, March 2017.
- 244 Bonnet, Céline and Schain, Jan Philip, An Empirical Analysis of Mergers: Efficiency Gains and Impact on Consumer Prices, February 2017.

- Benndorf, Volker and Martinez-Martinez, Ismael, Perturbed Best Response Dynamics in a Hawk-Dove Game, January 2017.
 Published in: Economics Letters, 153 (2017), pp. 61-64.
- 242 Dauth, Wolfgang, Findeisen, Sebastian and Suedekum, Jens, Trade and Manufacturing Jobs in Germany, January 2017. Forthcoming in: American Economic Review, Papers & Proceedings.
- 241 Borrs, Linda and Knauth, Florian, The Impact of Trade and Technology on Wage Components, December 2016.
- 240 Haucap, Justus, Heimeshoff, Ulrich and Siekmann, Manuel, Selling Gasoline as a By-Product: The Impact of Market Structure on Local Prices, December 2016.
- 239 Herr, Annika and Normann, Hans-Theo, How Much Priority Bonus Should be Given to Registered Organ Donors? An Experimental Analysis, November 2016.
- 238 Steffen, Nico, Optimal Tariffs and Firm Technology Choice: An Environmental Approach, November 2016.
- 237 Behrens, Kristian, Mion, Giordano, Murata, Yasusada and Suedekum, Jens, Distorted Monopolistic Competition, November 2016.
- 236 Beckmann, Klaus, Dewenter, Ralf and Thomas, Tobias, Can News Draw Blood? The Impact of Media Coverage on the Number and Severity of Terror Attacks, November 2016. Forthcoming in: Peace Economics, Peace Science and Public Policy.
- 235 Dewenter, Ralf, Dulleck, Uwe and Thomas, Tobias, Does the 4th Estate Deliver? Towars a More Direct Measure of Political Media Bias, November 2016.
- 234 Egger, Hartmut, Kreickemeier, Udo, Moser, Christoph and Wrona, Jens, Offshoring and Job Polarisation Between Firms, November 2016.
- 233 Moellers, Claudia, Stühmeier, Torben and Wenzel, Tobias, Search Costs in Concentrated Markets An Experimental Analysis, October 2016.
- 232 Moellers, Claudia, Reputation and Foreclosure with Vertical Integration Experimental Evidence, October 2016.
- 231 Alipranti, Maria, Mitrokostas, Evangelos and Petrakis, Emmanuel, Non-comparative and Comparative Advertising in Oligopolistic Markets, October 2016. Forthcoming in: The Manchester School.
- 230 Jeitschko, Thomas D., Liu, Ting and Wang, Tao, Information Acquisition, Signaling and Learning in Duopoly, October 2016.
- 229 Stiebale, Joel and Vencappa, Dev, Acquisitions, Markups, Efficiency, and Product Quality: Evidende from India, October 2016.
- 228 Dewenter, Ralf and Heimeshoff, Ulrich, Predicting Advertising Volumes: A Structural Time Series Approach, October 2016. Published in: Economics Bulletin, 37 (2017), Volume 3.
- 227 Wagner, Valentin, Seeking Risk or Answering Smart? Framing in Elementary Schools, October 2016.
- Moellers, Claudia, Normann, Hans-Theo and Snyder, Christopher M., Communication in Vertical Markets: Experimental Evidence, July 2016.
 Published in: International Journal of Industrial Organization, 50 (2017), pp. 214-258.

- 225 Argentesi, Elena, Buccirossi, Paolo, Cervone, Roberto, Duso, Tomaso and Marrazzo, Alessia, The Effect of Retail Mergers on Prices and Variety: An Ex-post Evaluation, June 2016.
- 224 Aghadadashli, Hamid, Dertwinkel-Kalt, Markus and Wey, Christian, The Nash Bargaining Solution in Vertical Relations With Linear Input Prices, June 2016. Published in: Economics Letters, 145 (2016), pp. 291-294.
- 223 Fan, Ying, Kühn, Kai-Uwe and Lafontaine, Francine, Financial Constraints and Moral Hazard: The Case of Franchising, June 2016. Forthcoming in: Journal of Political Economy.
- 222 Benndorf, Volker, Martinez-Martinez, Ismael and Normann, Hans-Theo, Equilibrium Selection with Coupled Populations in Hawk-Dove Games: Theory and Experiment in Continuous Time, June 2016. Published in: Journal of Economic Theory, 165 (2016), pp. 472-486.
- 221 Lange, Mirjam R. J. and Saric, Amela, Substitution between Fixed, Mobile, and Voice over IP Telephony – Evidence from the European Union, May 2016. Published in: Telecommunications Policy, 40 (2016), pp. 1007-1019.
- 220 Dewenter, Ralf, Heimeshoff, Ulrich and Lüth, Hendrik, The Impact of the Market Transparency Unit for Fuels on Gasoline Prices in Germany, May 2016. Published in: Applied Economics Letters, 24 (2017), pp. 302-305.
- 219 Schain, Jan Philip and Stiebale, Joel, Innovation, Institutional Ownership, and Financial Constraints, April 2016.
- 218 Haucap, Justus and Stiebale, Joel, How Mergers Affect Innovation: Theory and Evidence from the Pharmaceutical Industry, April 2016.
- 217 Dertwinkel-Kalt, Markus and Wey, Christian, Evidence Production in Merger Control: The Role of Remedies, March 2016.
- 216 Dertwinkel-Kalt, Markus, Köhler, Katrin, Lange, Mirjam R. J. and Wenzel, Tobias, Demand Shifts Due to Salience Effects: Experimental Evidence, March 2016. Published in: Journal of the European Economic Association, 15 (2017), pp. 626-653.
- 215 Dewenter, Ralf, Heimeshoff, Ulrich and Thomas, Tobias, Media Coverage and Car Manufacturers' Sales, March 2016. Published in: Economics Bulletin, 36 (2016), pp. 976-982.
- 214 Dertwinkel-Kalt, Markus and Riener, Gerhard, A First Test of Focusing Theory, February 2016.
- 213 Heinz, Matthias, Normann, Hans-Theo and Rau, Holger A., How Competitiveness May Cause a Gender Wage Gap: Experimental Evidence, February 2016. Forthcoming in: European Economic Review, 90 (2016), pp. 336-349.
- 212 Fudickar, Roman, Hottenrott, Hanna and Lawson, Cornelia, What's the Price of Consulting? Effects of Public and Private Sector Consulting on Academic Research, February 2016.
- 211 Stühmeier, Torben, Competition and Corporate Control in Partial Ownership Acquisitions, February 2016. Published in: Journal of Industry, Competition and Trade, 16 (2016), pp. 297-308.
- 210 Muck, Johannes, Tariff-Mediated Network Effects with Incompletely Informed Consumers, January 2016.

- 209 Dertwinkel-Kalt, Markus and Wey, Christian, Structural Remedies as a Signalling Device, January 2016.
 Published in: Information Economics and Policy, 35 (2016), pp. 1-6.
- 208 Herr, Annika and Hottenrott, Hanna, Higher Prices, Higher Quality? Evidence From German Nursing Homes, January 2016. Published in: Health Policy, 120 (2016), pp. 179-189.
- 207 Gaudin, Germain and Mantzari, Despoina, Margin Squeeze: An Above-Cost Predatory Pricing Approach, January 2016. Published in: Journal of Competition Law & Economics, 12 (2016), pp. 151-179.
- 206 Hottenrott, Hanna, Rexhäuser, Sascha and Veugelers, Reinhilde, Organisational Change and the Productivity Effects of Green Technology Adoption, January 2016. Published in: Energy and Ressource Economics, 43 (2016), pp. 172–194.
- 205 Dauth, Wolfgang, Findeisen, Sebastian and Suedekum, Jens, Adjusting to Globalization – Evidence from Worker-Establishment Matches in Germany, January 2016.
- 204 Banerjee, Debosree, Ibañez, Marcela, Riener, Gerhard and Wollni, Meike, Volunteering to Take on Power: Experimental Evidence from Matrilineal and Patriarchal Societies in India, November 2015.
- 203 Wagner, Valentin and Riener, Gerhard, Peers or Parents? On Non-Monetary Incentives in Schools, November 2015.
- 202 Gaudin, Germain, Pass-Through, Vertical Contracts, and Bargains, November 2015. Published in: Economics Letters, 139 (2016), pp. 1-4.
- 201 Demeulemeester, Sarah and Hottenrott, Hanna, R&D Subsidies and Firms' Cost of Debt, November 2015.
- 200 Kreickemeier, Udo and Wrona, Jens, Two-Way Migration Between Similar Countries, October 2015. Forthcoming in: World Economy.
- Haucap, Justus and Stühmeier, Torben, Competition and Antitrust in Internet Markets, October 2015.
 Published in: Bauer, J. and M. Latzer (Eds.), Handbook on the Economics of the Internet, Edward Elgar: Cheltenham 2016, pp. 183-210.
- 198 Alipranti, Maria, Milliou, Chrysovalantou and Petrakis, Emmanuel, On Vertical Relations and the Timing of Technology, October 2015. Published in: Journal of Economic Behavior and Organization, 120 (2015), pp. 117-129.
- 197 Kellner, Christian, Reinstein, David and Riener, Gerhard, Stochastic Income and Conditional Generosity, October 2015.
- 196 Chlaß, Nadine and Riener, Gerhard, Lying, Spying, Sabotaging: Procedures and Consequences, September 2015.
- 195 Gaudin, Germain, Vertical Bargaining and Retail Competition: What Drives Countervailing Power?, May 2017 (First Version September 2015). Forthcoming in: The Economic Journal.
- 194 Baumann, Florian and Friehe, Tim, Learning-by-Doing in Torts: Liability and Information About Accident Technology, September 2015.

- 193 Defever, Fabrice, Fischer, Christian and Suedekum, Jens, Relational Contracts and Supplier Turnover in the Global Economy, August 2015. Published in: Journal of International Economics, 103 (2016), pp. 147-165.
- 192 Gu, Yiquan and Wenzel, Tobias, Putting on a Tight Leash and Levelling Playing Field: An Experiment in Strategic Obfuscation and Consumer Protection, July 2015. Published in: International Journal of Industrial Organization, 42 (2015), pp. 120-128.
- 191 Ciani, Andrea and Bartoli, Francesca, Export Quality Upgrading under Credit Constraints, July 2015.
- 190 Hasnas, Irina and Wey, Christian, Full Versus Partial Collusion among Brands and Private Label Producers, July 2015.
- Dertwinkel-Kalt, Markus and Köster, Mats, Violations of First-Order Stochastic Dominance as Salience Effects, June 2015.
 Published in: Journal of Behavioral and Experimental Economics, 59 (2015), pp. 42-46.
- 188 Kholodilin, Konstantin, Kolmer, Christian, Thomas, Tobias and Ulbricht, Dirk, Asymmetric Perceptions of the Economy: Media, Firms, Consumers, and Experts, June 2015.
- 187 Dertwinkel-Kalt, Markus and Wey, Christian, Merger Remedies in Oligopoly under a Consumer Welfare Standard, June 2015. Published in: Journal of Law, Economics, & Organization, 32 (2016), pp. 150-179.
- 186 Dertwinkel-Kalt, Markus, Salience and Health Campaigns, May 2015. Published in: Forum for Health Economics & Policy, 19 (2016), pp. 1-22.
- 185 Wrona, Jens, Border Effects without Borders: What Divides Japan's Internal Trade? May 2015.
- Amess, Kevin, Stiebale, Joel and Wright, Mike, The Impact of Private Equity on Firms' Innovation Activity, April 2015.
 Published in: European Economic Review, 86 (2016), pp. 147-160.
- 183 Ibañez, Marcela, Rai, Ashok and Riener, Gerhard, Sorting Through Affirmative Action: Three Field Experiments in Colombia, April 2015.
- 182 Baumann, Florian, Friehe, Tim and Rasch, Alexander, The Influence of Product Liability on Vertical Product Differentiation, April 2015. Published in: Economics Letters, 147 (2016), pp. 55-58 under the title "Why Product Liability May Lower Product Safety".
- 181 Baumann, Florian and Friehe, Tim, Proof beyond a Reasonable Doubt: Laboratory Evidence, March 2015.
- 180 Rasch, Alexander and Waibel, Christian, What Drives Fraud in a Credence Goods Market? – Evidence from a Field Study, March 2015.
- 179 Jeitschko, Thomas D., Incongruities of Real and Intellectual Property: Economic Concerns in Patent Policy and Practice, February 2015. Forthcoming in: Michigan State Law Review.
- 178 Buchwald, Achim and Hottenrott, Hanna, Women on the Board and Executive Duration Evidence for European Listed Firms, February 2015.
- 177 Heblich, Stephan, Lameli, Alfred and Riener, Gerhard, Regional Accents on Individual Economic Behavior: A Lab Experiment on Linguistic Performance, Cognitive Ratings and Economic Decisions, February 2015. Published in: PLoS ONE, 10 (2015), e0113475.

- 176 Herr, Annika, Nguyen, Thu-Van and Schmitz, Hendrik, Does Quality Disclosure Improve Quality? Responses to the Introduction of Nursing Home Report Cards in Germany, February 2015. Published in: Health Policy, 120 (2016), pp.1162-1170.
- Herr, Annika and Normann, Hans-Theo, Organ Donation in the Lab: Preferences and Votes on the Priority Rule, February 2015.
 Published in: Journal of Economic Behavior and Organization, 131 Part B (2016), pp. 139-149.
- 174 Buchwald, Achim, Competition, Outside Directors and Executive Turnover: Implications for Corporate Governance in the EU, February 2015.
- 173 Buchwald, Achim and Thorwarth, Susanne, Outside Directors on the Board, Competition and Innovation, February 2015.
- 172 Dewenter, Ralf and Giessing, Leonie, The Effects of Elite Sports Participation on Later Job Success, February 2015.
- 171 Haucap, Justus, Heimeshoff, Ulrich and Siekmann, Manuel, Price Dispersion and Station Heterogeneity on German Retail Gasoline Markets, January 2015. Forthcoming in: The Energy Journal.
- 170 Schweinberger, Albert G. and Suedekum, Jens, De-Industrialisation and Entrepreneurship under Monopolistic Competition, January 2015. Published in: Oxford Economic Papers, 67 (2015), pp. 1174-1185.

Older discussion papers can be found online at: <u>http://ideas.repec.org/s/zbw/dicedp.html</u>

Heinrich-Heine-University of Düsseldorf

Düsseldorf Institute for Competition Economics (DICE)

Universitätsstraße 1_40225 Düsseldorf www.dice.hhu.de